ELSEVIER

Contents lists available at ScienceDirect

# Energy Research & Social Science

journal homepage: www.elsevier.com/locate/erss





# Top-down sustainability transitions in action: How do incumbent actors drive electric mobility diffusion in China, Japan, and California?

Gregory Trencher a,b,\*, Nhi Truong b, Pinar Temocin c, Mert Duygan d,e

- <sup>a</sup> Graduate School of Global Environmental Studies, Kyoto University, Kyoto, Japan
- <sup>b</sup> Graduate School of Environmental Studies, Tohoku University, Miyagi Prefecture, Japan
- <sup>c</sup> Graduate School for International Development and Cooperation, Hiroshima University, Hiroshima Prefecture, Japan
- <sup>d</sup> Faculty of Business and Economics, University of Basel, Switzerland
- e Department of Environmental Social Sciences, Swiss Federal Institute of Aquatic Science and Technology (Eawag), Switzerland

#### ARTICLE INFO

#### Keywords: Sustainability transitions Electric vehicles Top-down Incumbent State Acceleration strategies

#### ABSTRACT

In explaining how socio-technical transitions occur, prevailing theories focus on bottom-up processes driven by new entrants, diverse actors and open-ended exploration in small, protected niches. Incumbent firms are frequently portrayed as hampering change, while managerial strategies using traditional public policy instruments remain understudied. Addressing this bias, we examine strategies used by networks of incumbent state and industry actors in China, Japan and California to accelerate the production and diffusion of battery-electric or hydrogen-powered vehicles. We build a comprehensive framework that systematically marries mechanisms of industrial transformation described in developmental-state literature with theories of socio-technical change from transitions scholarship. We then use a vast dataset of secondary documents and interviews to examine the principal strategies employed in each country, identifying variations over two phases of technological diffusion. Findings reveal that the incumbent actor networks in each country have collectively employed multiple but similar strategies. Yet closer inspection of specific policy instruments, such as regulations and performance-based incentives, along with ambitions to phase out vehicles with internal combustion engines, reveals differences across cases. We explain these by considering different motivations for each country's transition and influencing socio-political conditions. Our study contributes to the enrichment of future transitions research in at least two ways. Theoretically, by integrating literature on transitions and developmental states, we deepen understanding of how incumbent state and market actors can attempt to drive socio-technical change. Empirically, our analysis provides important evidence for understanding the strategies driving top-down transitions outside northern Europe, and the conditions affecting instrument choice.

#### 1. Introduction

Scholars are devoting increasing attention to socio-technical transitions driven by incumbent industry and elite state actors, such as government ministries and political leaders [1,2]. With innovation activities by established firms often featuring in literature [3–5], interest is mounting in the role of state actors [6,7] and policy instruments in the pursuit of 'politically accelerated' transitions [8–10]. Scholars highlight the need for state intervention not only for spurring the creation and diffusion of new technologies and industries, but equally for hastening the decline of carbon-intensive energy systems and other environmentally harmful technologies, processes, and substances [11,12]. Thus, in addition to newcomers challenging the status quo, incumbent

government and market actors are potentially powerful drivers of sociotechnical transitions [9,10,13,14].

Incumbents in government and industry typically possess vast financial, human, and intellectual capital and direct or strong influence on markets, policies, and politics [15]. This power and resources lead naturally to top-down approaches [10]. These typically involve managerial strategies built around specific objectives and timelines, highly-guided search paths, aggressive intervention in markets through regulation or economic incentives, and large-scale investments in infrastructure [16]. The strength of top-down approaches driven by powerful incumbents lies in the potential to trigger widescale socio-technical change within a short timeframe due to the concentration of resources and agency [17,18]. However, by overlooking local circumstances and

<sup>\*</sup> Corresponding author at: Graduate School of Environmental Studies, Tohoku University, Miyagi Prefecture, Japan.

needs while sidestepping inclusiveness for the sake of scale, speed and efficiency, top-down blueprints for change run the risk of meeting resistance from the interests they underrepresent [19,20].

The idea of top-down, highly coordinated strategies orchestrated by incumbents in government and industry contrasts with the dominant theories. Transitions scholarship has historically viewed socio-technical change as resulting organically from bottom-up approaches [21]. In this paradigm, innovation is driven by newcomer firms and inclusive governance networks linking citizens, scientists and policymakers [22-24]. These diverse actors collaborate in creative visioning and open-ended experimentation in small-scale, protected 'niches'. Change is purported to occur during windows of opportunity, as the collective influence of 'regimes' (established webs of technologies, actors and institutions) is momentarily weakened by the simultaneous emergence of new technologies or business models and 'landscape' pressures from markets or society [12,13,16,24]. While bottom-up approaches are attractive by virtue of their inclusiveness and attachment to specific communities and needs [25], their ability to rapidly bring about widespread change may be sometimes limited.

Nevertheless, numerous studies by transitions scholars implicitly advocate bottom-up approaches by portraying incumbent actors as hindering innovation [15,26–28] and wedded to unsustainable sociotechnical configurations and business practices. Difficulties in breaking with established practices are often seen to result from the lock-in effect of pre-existing assets, institutions, and financial or intellectual resources [29–33]. Several studies [34,35] also describe incumbent firms as actively resisting change via strategies like political lobbying, shared discourse, and renewed investment in current technologies.

Transition scholars have recently highlighted several limitations in the explanatory power of views of change biased towards bottom-up approaches [1,23,28]. First, literature has tended to overlook many instances where incumbent firms drive technological or business innovation and experimentation in protected niches [3,16,26]. Second, accounts of state actors driving transitions [6,8,10,27,36,37] and the influence of conventional policy instruments such as planning, regulation, and public investments [15,21] have received limited attention. Third, some scholars [2,38] claim that literature's emphasis on nichedriven innovation and participatory governance reflects a normative, Western bias in socio-political conceptualisations of how socio-technical transitions occur. Moreover, not only does the emphasis on bottom-up approaches and newcomers contrast with governance norms in many non-Western settings such as Asia, but established theory is ill-suited to explaining how incumbent government and market actors have collaboratively driven socio-technical change in progressive green economies such as California, Germany, and Sweden [7,39,40].

Insights from developmental-state literature can overcome these tensions between prevailing theories and empirical exceptions. Rejecting the neoliberal idea that economic growth and innovation are best achieved in laissez-faire markets [41,42], literature on so-called 'developmental states' seeks to explain the causality between top-down state intervention and industrial, technological, or economic transformations [43,44]. A core goal of the development-state model is to drive economic growth through industrial policy targeted at specific sectors selected for state support [44,45]. Typically used policy instruments include long-term planning and target setting, subsidies for manufacturers, tight government–industry networks for information sharing, state-mediated collaborations across competing industries, public investments in human resources, R&D and infrastructure, and preferential treatment for domestic firms [41,46].

While this field has historically focused on catch-up-style economic development in East Asian nations like Japan, Korea, Singapore and Taiwan during the 1980s–1990s [47,48], recent scholarship has extended this theory of change to explain energy, mobility, and sustainability transitions in various Asian [17,42,49,50], Western [39,51] and developing countries [52]. The emerging field of the green

developmental state [53,54] thus provides an important opportunity to enrich transitions scholars' understanding of governance strategies that incumbent state and market actors can use to accelerate the supply and diffusion of green technologies, infrastructure, and socio-technical systems [6,36,45].

In this context, this paper aims to build theoretical and empirical understanding into the collaborative actions that networks of incumbent actors in government and industry can play in driving socio-technical transitions. Through a triple case study, our objective is to elucidate and compare the nature of strategies collectively used by communities of incumbents in China, Japan, and California to accelerate the production and diffusion of electric mobility (either plug-in electric or hydrogen) in different socio-political settings. Using primary and secondary data sourced from academic and grey literature along with interviews, we set out to answer: What principal strategies are collectively used by the incumbent actor network in each country? How and why do these strategies change? We also briefly consider outcomes achieved and limitations of strategies in each case.

Our contribution to literature is three-fold. First, we address the growing interest in top-down transitions driven by government intervention [18,55] and discussions about 'whether transitions can be engineered and planned from above' [7: 7]. With scholars calling for more studies on this topic [13], our novelty lies in our detailed empirics and comparative approach. Second, we marry two fields of literature: sustainability transitions and the developmental state. While transitions scholars are increasingly interested in the latter [6,51,52], rigorous treatment of the compatible or conflicting theories in each field has lacked until now. By providing an enriched perspective on how sociotechnical transitions can occur, this approach addresses calls to crossfertilise transitions theory with other fields [15,23]. Third, the framework developed to guide our analysis provides a finer-grained picture of a potential toolkit for steering sustainability transitions than previous conceptualisations [56-59]. Incorporating both the creative and destructive dimensions of socio-technical change [12,27,60,61], our framework provides a useful heuristic that other scholars might empirically apply to deepen understanding of what strategies can be used to steer sustainability transitions and how these might vary over different phases of technological development or socio-political contexts.

Henceforth, Section 2 will synthesise literature from sustainability transitions and the developmental state into a framework of governance strategies deemed crucial for accelerating socio-technical change. Section 3 presents our study design and unpacks the methods underpinning our case analysis. Section 4 introduces the cases. Section 5 applies our framework to compare the electric mobility transition strategies of China, Japan, and California. Section 6 summarises key findings and clarifies key characteristics in each case. Finally, Section 7 identifies theoretical implications for scholarship.

# 2. Analytical framework: governance strategies for accelerating sustainability transitions

# 2.1. Overview of the literature review

For literature on sustainability transitions, we include empirical and theoretical studies from its four subfields [see 15, 62]: multi-level perspective, strategic niche management, transitions management, and technological innovation systems. Relevant studies were identified by searching Web of Science with keyword combinations and examining their references and citation networks. We also consulted two policymaker reports by transition scholars [63,64]. In selecting papers for analysis, we prioritised those with explicit descriptions of strategies

<sup>1 &#</sup>x27;Socio-technical', 'transition', 'governance', 'strategies', 'acceleration', 'manage', 'success', 'drivers', 'levers', 'conditions', 'sustainability', 'green', 'energy', 'mobility', 'transport', 'technology', etc.

seen as important for accelerating socio-technical transitions.

For literature on the developmental state, we extracted the quintessential functions of this governance paradigm from classic works of the 1990s [41,43,44,47,48]. We then searched Web of Science with 'developmental state' and combinations of 'sustainability', 'green', 'low-carbon', 'environment', 'transition', and 'energy' to identify recent work from an environmental perspective. We also reviewed several studies outside this field that discuss the state's role in environmental governance [e.g. 39].

# 2.2. Explanation of the framework

The resulting analytical framework appears in Table 1. This shows generic strategies and specific instruments to implement these (e.g. policies and institutional arrangements). We henceforth follow a two-layered coding system, using C1–8 to denote *categories* of strategies and S1.1–8.4 for *individual* strategies.

The listed strategies were identified both deductively and inductively. We started with articulations provided in specific studies, finetuning these to reflect the larger plurality of descriptions identified in both fields of literature. We acknowledge the abundance of existing frameworks that describe various tactics for accelerating socio-technical transitions [56–59,65,66]. Duly integrating this scholarship, our interest in systematically comparing the governance strategies in multiple cases required a comprehensive and higher-resolution framework than found in existent literature.

The framework consists of three broad dimensions: (i) overarching coordination strategies, (ii) targeted strategies for accelerating creation and diffusion, and (iii) targeted strategies for accelerating decline. The first refers to wide-reaching strategies related to governance and management (e.g. overarching policies, technology roadmaps, governance bodies). The other two consist of strategies addressing more specific objectives and challenges in the goal of accelerating either the creation or decline of new technology and industries.

The framework is rooted in multiple literatures. First, it covers four key dimensions of technology policy: production, demand/market creation, infrastructure, and institutions [67-69]. Second, it encompasses both the creation of new sustainable technologies and the abolishment of environmentally harmful precedents [58,60,61,70]. For this, we draw largely on Kivimaa and Kern [57], who extend the basic governance functions suggested in technology innovation systems literature [56,71]. Yet our framework distinguishes itself from previous work. It integrates strategies widely emphasised in transitions or developmental-state literature that have not featured prominently in frameworks to date. These include inclusive governance (S1.2), monitoring and disclosure (S3.2), industry protection (S5.4), removing market barriers (S6.3), and sectoral coupling (S6.4). Third, by integrating insights into industrialchange mechanisms from developmental-state scholars, we explicitly integrate supply-side measures (e.g. S5.2-5.4), which are often underemphasised in transitions literature. In defining these strategies, we do not explicitly differentiate their suitability for top-down or bottom-up application, since they might be employed in either model [21,72].

Overall, we found that the bulk of strategies identified are seen by both fields of literature as important for accelerating socio-technical and industrial transformation. This is despite different theories of change and contrasting emphasis on variables like top-down *vs* bottom-up, incumbency *vs* inclusiveness/newcomers, open-ended experimentation *vs* managerial-type steering, and supply-side *vs* demand-side measures. However, since the emphasis placed on each strategy differs somewhat across time or the respective strand of literature, we discuss these variations below.

# 2.2.1. Knowledge management (C1)

Sustainability transitions and developmental-state scholars widely emphasise the importance of *knowledge dissemination* (S1.1). Frequently stressed is network creation to accelerate the identification and sharing

of technological breakthroughs, promising business and policy models and experiences accumulated by firms, knowledge producers, and government agencies. State actors are expected to play a key role by linking players and mediating across competing firms. Following Hekkert [56] and Meleen and Farla [66], we interpret this strategy as distinct but complementary to *knowledge production* (S5.3), where policy instruments such as R&D grants are more relevant.

#### 2.2.2. Actor networks and collaboration (C2)

Inclusive governance (S2.1) is widely normalised in transitions literature based on views that participation by diverse actors is vital to developing socially relevant change strategies [73,74]. Classic literature on developmental states does not share this emphasis. Recently, however, several scholars problematise this limitation, calling for diverse social inclusion - especially for tackling environmental issues [46,53,75]. In addition, state-industry collaboration (S2.2) is heavily prescribed in the developmental-state model. The participatory view historically advocated by many transition scholars underemphasises the need for linking incumbent government and market actors [76]. However, multiple studies have recently shown that incumbent industry involvement has played a major role in driving historical sustainability transitions [10,16,74]. Finally, while both strands voice a need for industry alliances (S2.3), each perspective differs. The 'David vs Goliath' view in transitions scholarship [77] has heavily emphasised the need for networks of newcomer firms to increase the collective ability to overcome resistance from incumbent opponents [55]. Yet less attention is paid to encouraging collaboration across incumbent firms to increase technological capabilities. In contrast, developmental-state literature heavily emphasises the state's role in coordinating the formation of alliances across competing incumbent firms [48,51,78].

### 2.2.3. Planning and commitment (C3)

Early transitions scholarship advocates for change through experimentation with open-ended visions and exploration [24]. Some recent studies, however, stress the importance of *planning and commitment* (S3.1) to influence the direction of innovation by fixing highly specific visions, goals and objectives [16,18,63]. This view concurs strongly with the managerial approach described by developmental-state scholars. Here, government targets for technology production or diffusion along with consistent long-term policy signals are seen as essential for giving confidence to technology producers to invest in new technologies and business models [41,44,49,50]. As stressed in both branches of literature, this 'blueprint' approach to transitions creates a need for *monitoring and disclosure* (S3.2) to enable the periodical feedback of results into policy and practice.

### 2.2.4. Legitimisation and advocacy (C4)

Given that new technologies often battle issues of unfamiliarity [79], transitions scholars frequently advocate the need to *influence public views* (S4.1) through outreach, education, and information dissemination [10,24]. Scholarship also underscores the need for *political lobbying* (S4.2) and advocacy coalitions aimed at boosting political support and favourable institutional arrangements and funding schemes [56]. Yet developmental-state literature places distinctly less emphasis on these strategies, likely for two reasons. First, by nature, the development-state model fosters the emergence of intimate cooperative relationships between policymakers and firms [17,80]. Second, the powerful political and marketing resources of incumbent firms and government agencies are well-suited to influencing public opinion [81].

#### 2.2.5. Industry creation and technology production (C5)

In advocating various strategies to spur the emergence of new industries and technologies, both strands of literature emphasise *cost alteration* (S5.1) through subsidies and pricing schemes [36,41,45,54,57,82]. Primarily concerned with supply-side strategies to boost the competitiveness of technology producers [54], the

 Table 1

 Framework of strategies for accelerating socio-technical transitions.

Strategy	Definition	Instrument examples	Transitions literature	Developmental state literature
tion strategies 1.1 Knowledge dissemination	Measures to support the broad dissemination of knowledge about the manufacturing or diffusion of new technologies (and associated policies, markets, business strategies and infrastructure, etc.).	Knowledge sharing networks, workshops and symposiums, R&D and innovation platforms, public–private organisations, technology transfer from universities or scientific institutions to industry, etc.	Hekkert, Suurs, Negro, Kuhlmann and Smits [56] Kivimaa and Kern [57] Jacobsson and Bergek [83] Sovacool and Martiskainen [74] Meelen and Farla [66] Smith, Stirling and Berkhout [65]	Öniş [41] Castells [47] MacNeil [90] Mazzucato [36] Chen and Lees [50] Swilling, Musango and Wakeford [52] Kronsell, Khan and Hildingsson [39]
2.1 Inclusive governance	Measures to include diverse or new actors into decision-making and governance frameworks to ensure that diverse viewpoints and priorities are reflected in technology, business and policy initiatives.	Decision-making frameworks such as advisory or steering committees, public debates etc. that include diverse actors (e.g. new entrants, ventures, NGOs, citizen groups).	Hoogma, Kemp, Schot and Truffer [91] Sovacool and Martiskainen [74] Geels, Turnheim, Asquith, Kern and Kivimaa [63] Johnstone, Rogge, Kivimaa, Fratini, Primmer and Stirling [92] Newell and Simms [7]	Angel and Rock [75] Lang, Wiek, Bergmann Stauffacher, Martens, Moll, Swilling and Thomas [94] Andrews and Shabani [95] Evans and Heller [96] Vazquez-Brust, Smith and Sarkis [53] Dent [49]
2.2 State-industry collaboration	Measures to link both government and incumbent industry actors into formal arrangements carrying out collaborative innovation and market development activities.	Public-private organisations, industry-government roadmaps and publications, public-private investments in infrastructure, R&D and market development, etc.	Ragerberg [93] Roberts, Geels, Lockwood, Newell, Schmitz, Turnheim and Jordan [13] Roberts and Geels [10] Turnheim and Geels [16] Cherp, Vinichenko, Jewell, Suzuki and Antal [14] Sovacool and Martiskainen [74] Harborne and Hendry [97]	Öniş [41] Johnson [44] Kronsell, Khan and Hildingsson [39] Lee, Jung and Lee [98 Kim [99] Meckling and Nahm [51] Kalinowski [80] Chien [54]
2.3 Industry alliances	Measures to link industry actors through formal partnerships or coalitions allowing like-minded firms to align activities and conduct joint technology creation, investments, business development, policy advice etc.	Joint investments in new businesses, R&D or infrastructure, formation of knowledge sharing and joint activity platforms, creation of industry organisations or networks, etc.	Roberts, Geels, Lockwood, Newell, Schmitz, Turnheim and Jordan [13] Lam, Martín-López, Wiek, Bennett, Frantzeskaki, Horcea- Milcu and Lang [100] Tzankova [101] Sovacool and Martiskainen [74] Turnheim and Geels [16] Geels, Turnheim, Asquith, Kern and Kivimaa [63] Hekkert, Suurs, Negro, Kuhlmann and Smits	Öniş [41] Scoones [78] Meckling [45] Meckling and Nahm [51] Andrews and Nwapi [46]
3.1 Planning and commitments	Measures to motivate actors, secure long-term commitments to a particular socio-technical trajectory and guide R&D, commercialisation and investments.	Visions, roadmaps and targets (e.g., for diffusion, cost and technical or environmental performance), planning documents, legal or voluntary commitments, etc.	[56] Rotmans, Kemp and van Asselt [24] Frank, Jacob and Quitzow [59] Smith, Stirling and Berkhout [65] Turnheim and Geels [16] Geels, Turnheim, Asquith, Kern and Kivimaa [63]	Öniş [41] Johnson [44] Wong [102] Kim [99] Chen and Lees [50] Mazzucato [36] Meckling and Nahm [51]
	tion strategies 1.1 Knowledge dissemination  2.1 Inclusive governance  2.2 State-industry collaboration  2.3 Industry alliances	tion strategies  1.1 Knowledge dissemination  Measures to support the broad dissemination of knowledge about the manufacturing or diffusion of new technologies (and associated policies, markets, business strategies and infrastructure, etc.).  2.1 Inclusive governance  Measures to include diverse or new actors into decision-making and governance frameworks to ensure that diverse viewpoints and priorities are reflected in technology, business and policy initiatives.  Measures to link both government and incumbent industry actors into formal arrangements carrying out collaborative innovation and market development activities.  Measures to link industry actors through formal partnerships or coalitions allowing like-minded firms to align activities and conduct joint technology creation, investments, business development, policy advice etc.  Measures to motivate actors, secure long-term commitments to a particular socio-technical trajectory and guide R&D, commercialisation and	tion strategies 1.1 Knowledge dissemination dissemination of knowledge about the manufacturing or diffusion of new technologies (and associated policies, markets, business strategies and infrastructure, etc.).  2.1 Inclusive governance  Measures to include diverse or new actors into decision-making and governance frameworks to ensure that diverse viewpoints and priorities are reflected in technology, business and policy initiatives.  Measures to link both government and incumbent industry actors into formal partnerships or coalitions allowing like-minded firms to align activities and conduct plint technology creation, investments, business development, policy advice etc.  3.1 Planning and commitments  Measures to motivate actors, secure long-term commitments on garticular socio-technical trajectory and guide R&D, commercialisation and RAD or infrastructure, formation of knowledge sharing networks, workshops and symposiums, R&D and innovation platfunnovation-private organisations, public-private organisations, industry-government roadmaps and public actions, public debase setc. that include diverse actors (e.g. new entrants, ventures, NGOs, citizen groups).  Public-private organisations, eventures, NGOs, citizen groups, advisor, public debase scit. that include diverse actors, eventures, NGOs, citizen groups, and advisory or steering committees, public debase scit. that include diverse actors, eventures, NGOs, citizen groups, and advisory	tion strategies 1.1 Knowledge dissemination  Measures to support the broad dissemination of Innowledge about the manufacturing or diffusion of new technologies (and associated policies, markets, business strategies and infrastructure, etc.).  2.1 Inclusive governance  2.2 Inclusive governance  Measures to include diverse or new actors into decision-making and governance frameworks to ensure that diverse everpoints and priorities are reflected in technology, business and policy initiatives.  Decision-making frameworks uch as advisory or steering committees diverse everpoints and priorities are reflected in technology, business and policy initiatives.  Measures to link hoth government and incumbent industry acrors into formal arrangements carrying out collaboration  Measures to link industry actors through formal partnerships or coalitions allowing like-minded firms to allajon activities and conduct ploit technology creation, investments, business development, policy advice etc.  Measures to link industry actors through formal partnerships or coalitions allowing like-minded firms to allajon activities and conduct ploit technology creation, investments, business development, policy advice etc.  Measures to link industry actors through formal partnerships or coalitions allowing like-minded firms to allajon activities and conduct ploit technology creation, investments in infrastructure, R8D and market development, policy advice etc.  Measures to link industry actors through formal partnerships or coalitions allowing like-minded firms to allajon activities and conduct ploit technology creation, investments in infrastructure, R8D and Antal 1141 Sovacoid and Antal 1

Table 1 (continued)

Category	Strategy	Definition	Instrument examples	Transitions literature	Developmental state literature
				Hekkert, Suurs, Negro, Kuhlmann and Smits [56]	
	3.2 Monitoring and disclosure	Measures to track and publicly disclose progress and learning from emerging experiences, to compare to initial goals and expectations, and to facilitate periodical reflection into policy or industry practices.	Progress reports, periodical disclosure of data, workshops, symposiums, etc.	Rotmans, Kemp and van Asselt [24] Geels, Turnheim, Asquith, Kern and Kivimaa [63] Markard [60] Sovacool and Martiskainen [74] Altenburg and Pegels [82]	Vazquez-Brust, Smitl and Sarkis [53] Scoones [78] Hughes [103]
	r accelerating creation		Dublic control of continuous	December Transferrence and	H F10/3
4. Legitimisation and advocacy	4.1 Influencing public views	Measures to positively influence public perceptions and acceptance of the technology (e.g. about benefits/risks, market growth, investment opportunities, etc.).	Public outreach, advertising and education, narratives about the need or desirability of technologies, public demonstrations, expositions, symposiums, etc.	Bergek, Jacobsson and Sandén [71] Rotmans, Kemp and van Asselt [24] Geels, Sovacool, Schwanen and Sorrell [79] Smith, Stirling and Berkhout [65] Nilsson and Nykvist [104] Roberts and Geels [10] Klitkou, Bolwig, Hansen and Wessberg	Han [106]
	4.2 Political	Measures to influence actions, plans or	Advocacy or lobbying coalitions,	[105] Hekkert, Suurs, Negro,	Kronsell, Khan and
	lobbying	decisions of government officials or organisations to facilitate the production or diffusion of technologies.	policy proposals, political donations, etc.	Kuhlmann and Smits [56] Bergek, Jacobsson and Sandén [71] Lowes, Woodman and Fitch-Roy [107] Kivimaa, Hyysalo, Boon, Klerkx, Martiskainen and Schot [108] Smith and Raven [85]	Hildingsson [39]
5. Industry creation and technology production	5.1 Cost alteration	Measures to improve the economic environment affecting the competitiveness of emerging technologies via policies that reduce the cost of production, purchase or use via economic incentives or market rules.	Pricing regulations (e.g. feed-in- tariffs or carbon/fossil-fuel taxes), voluntary price ceilings, tax waivers, performance incentives, etc.	Kivimaa and Kern [57] Roberts and Geels [10] Altenburg and Pegels [82] Nilsson and Nykvist [104] Geels, Turnheim, Asquith, Kern and Kivimaa [63] Jacobsson and Lauber [109]	Öniş [41] Wade [48] Mazzucato [36] Chien [54] Meckling [45] Eckersley [89]
	5.2 Industry nurturing	Measures to support the creation or expansion of new or larger technology producers to increase the supply of technology.	Tax credits, land provision, grants, finance provision, formation of technology parks or regional clusters, local content requirements, etc.	Quitzow [110] Klitkou, Bolwig, Hansen and Wessberg [105] Harborne and Hendry [97] Cooke [111]	Öniş [41] Kim [99] Wong [102] Chien [54] Chen and Lees [112] Meckling [45]
	5.3 Knowledge production	Measures that support the creation of new knowledge (either scientific or business-related) in firms, research institutes or government agencies.	R&D subsidies, fiscal incentives, formation of technology parks, joint knowledge production platforms, etc.	Kanger, Sovacool and Noorkōiv [58] Kivimaa and Kern [57] Frank, Jacob and Quitzow [59] Smith, Stirling and Berkhout [65] Hekkert, Suurs, Negro, Kuhlmann and Smits [56] Meelen and Farla [66]	Wade [48] Castells [47] Mazzucato [36] Swilling, Musango ar Wakeford [52] Block [113] Meckling [45]

Table 1 (continued)

Category	Strategy	Definition	Instrument examples	Transitions literature	Developmental state literature
	5.4 Industry protection	Measures to protect domestic industry (especially technology manufacturers) from overseas competition.	Import tariffs, discriminatory subsidy allocation, local procurement requirements, etc.	Mazur, Contestabile, Offer and Brandon [114]	Öniş [41] Wade [48] World Bank [84] Angel and Rock [75] Valentine and Sovacool [115] Andrews and Nwapi [46]
6. Market creation and technology diffusion	6.1 Niche creation and experimentation	Measures to enable or protect the demonstration, testing or upscaling of novel practices (e.g. technologies, policies, social arrangements and business models etc.) that are typically limited geographically or temporally.	Public demonstrations, pilot zones, innovation districts, living labs, etc.	Kivimaa and Kern [57] Geels, Turnheim, Asquith, Kern and Kivimaa [63] Frank, Jacob and Quitzow [59] Hekkert, Suurs, Negro, Kuhlmann and Smits [56] Bergek, Jacobsson and Sandén [71] Hoogma, Kemp, Schot and Truffer [91]	Swilling, Musango and Wakeford [52] Eckersley [89] Scoones [78]
	6.2 Demand creation	Measures that target technology users to increase or stimulate new purchases.	Monetary incentives (e.g. subsidies or tax reductions), non-monetary incentives (e.g. access to car pool lanes), public outreach/marketing, green product labelling, public procurement standards, demonstration projects, etc.	Turnheim and Geels [16] Victor, Geels and Sharpe [64] Nilsson, Hillman and Magnusson [116] Dijk, Iversen, Klitkou, Kemp, Bolwig, Borup and Møllgaard [67] Figenbaum [117]	Mazzucato [36] Meckling [45] Chien [54]
	6.3 Removing market barriers	Measures to modify or remove policies and formal institutions blocking or disincentivising investments in the production or usage of new technologies and businesses.	Establishment of uniform codes and standards for technology design or operation, removal or relaxing of outdated laws and regulations (e.g. safety standards), etc.	Smith and Raven [85] Victor, Geels and Sharpe [64] Geels, Turnheim, Asquith, Kern and Kivimaa [63] Meelen and Farla [66]	Wade [48] Johnson [44] Mathews [118] MacNeil [90]
	6.4 Sectoral coupling	Measures to link the production or usage of common technologies across different markets to accelerate technological improvement and cost reduction via economies of scale.	R&D subsidies, demonstration programs, roadmaps or target setting to stimulate the production or usage of common base technologies (e.g. fuel cells and batteries) or energies across multiple societal sectors.	Geels, Sovacool, Schwanen and Sorrell [79] Geels [23] Markard, Geels and Raven [55] Frank, Jacob and Quitzow [59]	None observed*
7. Resource mobilisation and allocation	7.1 Financial support	Measures that directly supply financial capital or revenue to support the production, demonstration or diffusion of new technologies and associated infrastructure or business models.	Purchase subsidies for consumers, grants and interest free loans or tax credits for industry, revenue generating market instruments (e.g. carbon credit trading), etc.	Smith, Stirling and Berkhout [65] Geels, Turnheim, Asquith, Kern and Kivimaa [63] Kivimaa and Kern [57] Newell and Simms [7] Hekkert, Suurs, Negro, Kuhlmann and Smits	Wade [48] Castells [47] World Bank [84] Mazzucato [36] Block [113] Chien [54] Meckling [45]
	7.2 Infrastructure preparation	Measures to establish the required infrastructure (e.g. smart grids, power storage and charging or refuelling stations) to support technology diffusion.	Infrastructure deployment schedules and targets, financial incentives for installation (e.g. grants, market incentives, tax reductions), building or town planning codes (e.g. mandating PEV chargers in new construction), etc.	Roberts and Geels [10] Nilsson and Nykvist [104] Valentine and Sovacool [119] Dijk, Iversen, Klitkou, Kemp, Bolwig, Borup and Møllgaard [67] Leibowicz [120]	Castells [47] Chien [54] Kim [121] Han [106] Eckersley [89] Block [113]
	7.3 Human resource development	Measures to accelerate the development of human resources to support emerging industries.	Training or employment grants, talent attraction incentives, establishing or investing in	Hekkert, Suurs, Negro, Kuhlmann and Smits [56] Victor, Geels and	Öniş [41] Castells [47] Chen and Lees [50] Vazquez-Brust, Smith
					(continued on next page)

Table 1 (continued)

Category	Strategy	Definition	Instrument examples	Transitions literature	Developmental state literature
Toward dentaria	o for accoloration dealine		educational/research institutions and technology parks, etc.	Sharpe [64] Kivimaa, Primmer and Lukkarinen [122] Kivimaa and Kern [57] Sovacool and Martiskainen [74]	and Sarkis [53] Block [113] Swilling, Musango and Wakeford [52]
incumbent cont	8.1 Technology control and restriction	Measures that explicitly aim to reduce or make the production, sale or usage of incumbent unsustainable technologies difficult, expensive or illegal.	Technology bans (for producers or users), production quotas, driving restrictions for ICE vehicles, rising environmental standards for emissions/energy efficiency/pollutants, environmental taxes, emissions trading, etc.	Kivimaa and Kern [57] Geels, Sovacool, Schwanen and Sorrell [79] Rosenbloom and Rinscheid [12] Kanger, Sovacool and Noorkõiv [58] Rosenbloom, Markard, Geels and Fuenfschilling [61] Markard [60]	Meckling and Nahm [51] Eckersley [89] Meckling [45] Vazquez-Brust, Smith and Sarkis [53]
	8.2 Support memoval existing support or incentives for incumbent, unsustainable technologies.  8.3 Structural and rule reforms megime rules (e.g. legislation, laws and standards) to foster new market entrants or new practices. Measures may not necessarily be directed at specific technologies or environmental impacts.	existing support or incentives for incumbent, unsustainable	Reduction or removal of subsidies and institutional or economic incentives and privileges (e.g. land provision, permits, contracts, tax benefits), etc.	Kivimaa and Kern [57] Turnheim and Geels [70] Geels, Sovacool, Schwanen and Sorrell [79] Rosenbloom and Rinscheid [12] Brauers, Oei and Walk [123]	Vazquez-Brust, Smith and Sarkis [53]
		Market and regulatory reforms such as market liberalisation, changes to public financing rules (e.g. regarding carbon-intensive investments), introduction of renewable energy portfolio standards or tax increases for technology sale/use (e.g. gasoline or vehicle taxes) etc.	Turnheim and Geels [70] Kivimaa and Kern [57] Geels, Turnheim, Asquith, Kern and Kivimaa [63] Smith and Raven [85]	Eckersley [89] Scoones [78]	
	8.4 Actor changes	Measures that alter the decision-making and institutional environment supporting incumbent unsustainable technologies by removing or rebuilding established actor bodies or organisations.	Reform or abolishment of actor composition in decision-making bodies, networks and institutional frameworks in government or industry.	Kivimaa and Kern [57] Normann [88] Cherp, Vinichenko, Jewell, Suzuki and Antal [14] Brauers, Oei and Walk [123] Roberts, Geels, Lockwood, Newell, Schmitz, Turnheim and Jordan [13]	Valentine [17]

developmental-state model actively pursues industry nurturing (S5.2) through policies that stimulate the scale and speed of technology production. Emphasis on supply-side strategies is less pronounced in transitions scholarship due to the focus on demand-side measures and technology diffusion. Studies on technological innovation systems are an exception, however. Both fields strongly recognise the need for government support for knowledge production (S5.3) [47,48,58,66,83], based on awareness that both newcomer and incumbent firms can lack financial resources to invest in R&D agendas with uncertain near-term reward. The classic developmental state also heavily stresses industry protection (S5.4) to shield fledgling industries, by giving local manufacturers preferential treatment (e.g. with subsidies) and discriminating against foreign competitors (e.g. via trade tariffs) [41,48,84]. While recent scholarship on the green developmental state emphasises orthodox measures like cost alteration (S5.1), interest in how to protect frail new industries from foreign markets remains high [45,46,50].

# 2.2.6. Market creation and technology diffusion (C6)

Both fields describe strategies to accelerate market creation and technology adoption. In developmental states, industrial change is driven by picking technological winners. There is thus low emphasis on niche creation and experimentation (S6.1) with novel technologies and social arrangements. Recently, however, some scholars integrate insights from sustainability transitions to emphasise open-ended experimentation with competing alternatives under guidance of the state [52]. As mentioned, demand creation (S6.2) tends not to feature in classic developmental-state literature due to overwhelming focus on supplyside policy and technology production for export. Yet recent scholarship, discussing the relevance of the developmental-state model for diffusing green technologies, underscores a need for demand-side policies [36,45]. As for removing market barriers (S6.3), lack of industry standardisation and outdated institutions such as laws and regulations can slow the diffusion of path-blazing technologies. Strategic niche management [85] and studies on electric mobility transitions [86],

though not widely stressed in transitions scholarship, are vocal about removing such market barriers. Conversely, this strategy is heavily emphasised in classic [48] and green developmental-state literature [87]. Finally, transitions literature often stresses the importance of *sectoral coupling* (S6.4) to stimulate the diffusion of mutually complementary base technologies across multiple industries [23,55]. Developmental-state literature does not prescribe this strategy, presumably due to the focus on sector-specific industrial policy.

#### 2.2.7. Resource mobilisation and allocation (C7)

Both fields recognise the need to mobilise financial, human, and infrastructural resources. Regarding *financial support* (S7.1), in the classic developmental-state model, subsidies to industry are distributed based on performance [48,50]. Both fields emphasise *infrastructure preparation* (S7.2) driven by state funding along with policies and institutional arrangements that stimulate private investments. Lastly, both fields frequently discuss *human resource development* (S7.3), emphasising instruments such as training schemes, employment incentives, and technology parks.

#### 2.2.8. Reducing incumbent technology production and use (C8)

Though neglected in early sustainability transitions literature, recent studies advocate strategies to accelerate the decline of environmentally harmful technologies [57,60,70,88] through technology control and restriction (S8.1) and support removal (S8.2). Emphasised instruments include bans or restrictions on the production or use of particular technologies, and removal of incentives like subsidies or institutional arrangements that support incumbent industry [12]. Similarly, 'destructive' policies, though absent from early developmental-state literature, have received attention in recent work examining state-led governance for climate change and green industrial transformation [45,51,53,89]. For structural and rule reforms (S8.3) and actor changes (S8.4), the developmental-state literature devotes thin attention to overhauling incumbent governance structures and actor bodies. Still, it is recognised that developmental states tend to foster path dependency and vested interests in the government-industry actor networks sharing allegiance to a particular technology [17].

As demonstrated above, the bulk of strategies for accelerating sociotechnical change are widely discussed in both fields of literature. A comprehensive framework has thus emerged to guide our comparison of how each country applies these in practice.

#### 3. Methods

This section briefly outlines key methodological details, focusing on our case selection and the data analysis procedure. The overall study design that guided our research procedure is summarised into Fig. 1.

#### 3.1. Conceptual scope, case selection and temporal focus

In choosing suitable cases for analysis, we set the following criteria, requiring that each feature:

- An ambitious, long-term plan to accelerate a transition to electric mobility:
- A prominent set of incumbent actors from government or industry driving the transition;
- A variety of governance strategies supported by government funding and policies;
- Ongoing and interconnected strategies extending from technological development and testing to market creation and diffusion.

Siding with recent literature [1], our conception of 'incumbent' actors in this study focuses mainly on powerful and established organisations in government and industry with a direct stake and influence over the socio-technical regime. By focusing on the collective strategies used

by such actor networks to deliberately accelerate socio-technical change, our view of incumbency differs somewhat from many studies that have by default assumed incumbent actors to possess 'vested interests in maintaining the status quo' and be focused on resisting rather than 'enabling transitions' [77, 124: 148, 125: 3].

Satisfying the above criteria and featuring such incumbent actors, China, Japan and California were chosen for analysis. Their principal features are summarised in Table 2.

Since this research aims to explore the nature of strategies used in varying contexts, the cases were also chosen to exhibit a contrasting representation of key variables. These include the: (i) prominence of incumbent industry in the main governance bodies and strategies used, (ii) geographical scale of the transition, (iii) political economy styles following variations described in literature [90,126], and (iv) technological focus: fuel-cell electric vehicles (FCEVs) or plug-in electric vehicles (PEVs); with the latter comprised of pure battery electric vehicles and plug-in hybrid variations. The cases vary sharply in these variables, representing diversity rather than typical examples [127].

Also satisfying these conditions, other countries such as Germany and Korea might be considered ripe candidates for analysis. Japan and California, however, were chosen so we could integrate knowledge and data obtained through previous research [68,128]. In addition, inclusion of our three cases is rationalised by China's global status as the largest PEV market (measured by on-road vehicle numbers) and by California and Japan similarly possessing the world's largest and second largest FCEV stock respectively [129,130]. Admittedly, the mix of national and sub-national scales poses challenges for comparison across cases. However, the geographical boundaries of our case selection followed the overall focus of market descriptions given in existent literature on each country [68,131–133]. We also focus on the scale where the activities of each country's incumbent actor network are most prevalent.

To build understanding of how top-down transition strategies evolve, we examine two key phases in the technological lifecycle. As guidance, we adopt the distinction of evolving phases of technological diffusion described in literature. Since these conceptions vary across transitions scholarship [24,134], we began by considering three key phases suggested by Kivimaa et al. [108] and Kanger and Schot [135]: predevelopment/start-up, acceleration, and stabilisation. This study examines the first two, given the early nature of the transitions discussed.

The first phase, pre-development and exploration, stretches from initial development, experimentation, testing, and demonstration of technologies to early commercialisation and diffusion. The second phase, acceleration, involves efforts to diffuse technologies, stimulate demand, and accelerate market penetration while continuing to support technological improvement. As Table 2 shows, the duration of phase two is set uniformly across cases, while phase one in California and Japan starts before China's due to the earlier nature of technological development and testing (see Section 4).

#### 3.2. Data sources and analysis

This study uses data derived from both primary sources (semi-structured interviews) and secondary sources (academic and grey literature). Although interview data helps deepen the analysis, we consider only secondary documents when summarising and comparing the governance strategies used in each country. This increases the replicability and transparency of our conclusions. Documents used as evidence are cited directly in our findings and in online supplementary material that provides more detailed explanations of strategies, summarised in Tables 3–5.

Details of interviews appear in Appendix A. Respondents for each country include incumbent actors such as government agencies, automakers and fuel suppliers along with experts in universities, private firms, and think tanks. Altogether, we drew on 44 interviews conducted for this and previous research [68,128]. All interviews shared the goal of building understanding of strategies used to spur the production and

# 1. Literature review and theory building

#### **Guiding question:**

- What strategies for accelerating socio-technical change and industrial transformation are emphasised in literature?
- Locate relevant studies in sustainability transitions and developmental state literature
- Identify descriptions of transition accelerating strategies
- Organise into framework
  - → See Table 1

# 2. Case selection and data collection

- Fix criteria and select relevant cases
  - → See Section 3.1 and Table 2
- Collect evidence of strategy use from secondary documents and interviews
  - → See Appendix A

# 3. Case analysis

#### **Guiding questions:**

- What strategies and instruments were used or underused in each case?
- How has the use of strategies evolved across phases?
- Code documents describing strategy use in accord with framework, using software (MAXQDA)
   → See Tables 3 - 5 and supplementary material
- Integrate interview data to enrich explanations

#### 4. Meta analysis

#### **Guiding questions**

- What patterns can be seen across cases in each phase?
- What contextual factors influenced transition approaches and use of specific instruments?
- Identify trends regarding strategy use across cases
   → See Section 6.1 and Table 6
- Identify patterns regarding use of specific governance instruments and influencing factors
   → See Section 6.2 and Table 7

Fig. 1. Summary of research design.

**Table 2** Principal attributes of three cases.

Country	China	Japan	California
Principal governance actors	National and sub- national governments (i.e. provincial and city)	State government and incumbent industry	State government and incumbent industry
Scale of transition	Country-wide	Country-wide	State-wide
Type of political economy	Authoritarian and highly coordinated market economy	Corporatist and highly coordinated market economy	Liberal and weakly coordinated market economy
Technological focus	Plug-in electric vehicles (PEVs)	Fuel-cell electric vehicles (FCEVs)	Fuel-cell electric vehicles (FCEVs)
Phase one: Pre- development and exploration	2006 – 2014	1999 – 2014	1999 and 2014
Phase two: Acceleration	2015 – 2020	2015 – 2020	2015 – 2020

diffusion of PEVs or FCEVs in each country, outcomes and challenges, and factors influencing instrument selection.

When summarising each country's transition approach (see Tables 3–5), we assign two values ('strong' and 'weak/not observed') to reflect how strongly a strategy is used. 'Strong' means a strategy is visible in multiple examples (at least two) and used consistently over multiple years (at least two), based on available evidence. Indicators in Appendix B guided our scoring. Values were cross-checked by two authors to minimise subjective interpretation. Though other studies [67,136] use finer-grained evaluations of policy-use intensity (e.g. weak/medium/strong), we adopted the simpler binary system due to the volume and heterogeneity of strategies examined and the difficulties of creating quantitative depictions for which objective evidence could be

# collected.

# 4. Introduction to case studies

To contextualise each country's mobility transition, the following sections outline the timelines of the two phases, key governance actors, and motivating factors. Since civil society is underrepresented in the principal governance body promoting each vehicle technology examined,<sup>2</sup> we focus actor descriptions on state and market actors.

#### 4.1. Battery electric mobility in China

The Chinese government supports the development and diffusion of battery, fuel cell, and hybrid powertrains under its New Energy Vehicles (NEVs) policy. Following the primary focus of existent literature [133,137] and government policy [138], our case focuses principally on PEVs (comprised of pure battery electric or plug-in hybrids, including passenger vehicles and buses). In line with literature [139], we selected the period  $2006-2014^3$  as phase one (pre-development and exploration) and 2015-2020 as phase two (acceleration). Around the start of phase one, national-government policy support for developing China's NEV industry as a strategic priority was strengthened through five-year plans [140]. Also during phase one, diffusion efforts began, focusing on

<sup>&</sup>lt;sup>2</sup> More diverse actor inclusion may occur in other governance frameworks with relevance to electric mobility in each country. This argument is based on the membership of the California Fuel Cell Partnership, Japan's Strategic Hydrogen and Fuel Cell Strategy Council, and literature cited for inclusive governance (S2.1) in Tables 3 and 4 for China and Japan.

<sup>&</sup>lt;sup>3</sup> In setting our temporal periods for analysis, each phase extends from January in the first year to December in the last year.

Table 3
China: Summary of strategies used.

China: Summary of stra	<u> </u>	Dhace	1. Dre-development and evaluration (2004, 2014)	Dhace	2: Acceleration (2015–2020)
Catagori	Chunkas		1: Pre-development and exploration (2006–2014)		
Category	Strategy	Use*	Instrument examples	Use*	Instrument examples
Overarching coordinatio  1. Knowledge management	on strategies 1.1 Knowledge dissemination	•	Establishing non-profit China Electric Vehicle Association to promote knowledge sharing across industry and government. Establishing industry alliances of state-owned automakers and part suppliers described in S2.3.	•	Instruments from phase one (see S1.1) continued. Establishing industry thinktank China EV100 to diffuse knowledge across industry, government and academia via meetings, publications and forums.
2. Actor networks and collaboration	2.1 Inclusive governance	$\bigcirc$	Use of this strategy not observed [139,145,167].	$\bigcirc$	Situation unchanged from phase one (see S2.1) [139,144,150].
	2.2 State–industry collaboration	•	Using state-owned enterprises to synchronise government ambitions and industry activity [145,147].  Participation of government actors in industry alliances [145].	•	Instruments from phase one (see S2.2) continued [151].
	2.3 Industry alliances	•	Obliging foreign automakers to partner with domestic firms via joint-ventures when producing vehicles in China [144].  Establishing industry alliances of state-owned automakers and part suppliers (e.g. State-Owned Electric Vehicle Industry Alliance) to promote knowledge sharing, joint research and business ventures [145.173].	•	Instruments from phase one continued (see S2.3). Promoting and establishing mergers (e.g. between FAW, Dongfeng and Changan) and joint-ventures (e.g. T3 Mobility) between state-owned enterprises and private firms [174].
3. Guidance and planning	3.1 Planning and commitments	•	Periodically signalling long-term government commitments to the NEV industry and priority areas via multi-year plans [132]. Setting numerical targets for NEV sales, infrastructure deployment and technical performance	•	Instruments from phase one continued (see S3.1) [163].
	3.2 Monitoring and disclosure	$\bigcirc$	[172]. Weak use of this strategy [175,176].		Situation unchanged from phase one (see S3.2) [150]. Data on PEV sales and infrastructure deployment is published disparately by various industry associations. Yet government data is not published in an integrated, consistent and annual format.
Targeted strategies for a 4. Legitimisation	ccelerating creation and 4.1 Influencing public views	diffusion	Demonstrating NEVs at high-profile international events (e.g. Beijing Summer Olympics in 2008). Introducing PEVs into public bus and taxi fleets via national pilot project programme (Ten Cities	•	Instruments from phase one (see S4.1) continued, namely the Shanghai International Automobile City via demonstrations of PEVs and intelligent connected vehicles [178].
			Thousand Vehicles) [138]. Establishing public demonstration, marketing and business development zones (e.g. NEV testing zones in Shanghai International Automobile City) [177].		
	4.2 Political lobbying	$\bigcirc$	Weak use of this strategy. No dedicated coalition for political lobbying observed.	$\bigcirc$	Situation unchanged from phase one (see S4.2).
5. Industry creation and technology	5.1 Cost alteration		Providing government subsidies to automakers to reduce production and sales costs [144,163].		Instruments from phase one (see S5.1) continued [163,165].
production			Providing government subsidies to pilot cities to reduce burden of PEV fleet purchases and installing charging infrastructure [132].		Setting national price ceiling for recharging rates in public and private stations [179,180].
	5.2 Industry nurturing		Providing R&D funds to support development of new or existing firms [181].		Instruments from phase one (see S5.2) continued [146,148,163,165].
			Requiring foreign firms to transfer NEV-related knowledge to domestic partner when forming joint- ventures [144]. Providing land, tax breaks and low-interest loans to		Mandating minimum shares of NEVs in public fleets to support production through increased demand [166].
	5.3 Knowledge production	•	support establishment of PEV factories [146,148]. Providing R&D funds (via '863' plan) to support development of core PEV technologies [143,182]. Establishing innovation clusters and research institutes [139,181]. Requiring foreign firms to transfer NEV-related	•	Instruments from phase one (see S5.3) continued, notably through R&D fund provision via 'National Key Research & Development Program' plan [183].
	5.4 Industry protection	•	knowledge to domestic partner when forming joint-ventures [144].  Setting import tariffs to increase sales price of imported vehicles [184].  Proferentially allocating government subsidies to	•	Instruments from phase one (see S5.4) continued, but weakened towards end of phase two [144,172].
6. Market creation and technology diffusion	6.1 Niche creation and experimentation	•	Preferentially allocating government subsidies to domestic auto/battery makers [172].  Establishing pilot cities programme to facilitate experimentation with PEVs in public fleets and novel policies or business models (e.g. battery swapping)	•	Instruments from phase one (see S6.1) continued [142,163].
	6.2 Demand creation		[141,167,171]. Setting mandatory procurement targets for government fleets [132,144].		

Table 3 (continued)

		Phase	1: Pre-development and exploration (2006–2014)	Phase	2: Acceleration (2015–2020)
Category	Strategy	Use*	Instrument examples	Use*	Instrument examples
			Reducing purchase prices via subsidies to manufacturers [180].  Providing driving privileges to PEV purchasers (e.g. exception from driving or licence plate restrictions) [139].		Instruments from phase one (see S6.2) continued [150,165].
	6.3 Removing market barriers	$\bigcirc$	Weak use of this strategy. Limited evidence of efforts to set national standards for PEV components and recharging infrastructure [140,171].	•	Setting national standards for charging interfaces, piles, data communication protocols and safety standards for batteries [179].  Streamlining or removing permitting procedures to accelerate the speed of installing charging infrastructure.
	6.4 Sectoral coupling	$\bigcirc$	Weak use of this strategy. Limited evidence of national efforts to couple PEV deployment with complementary technologies (e.g. smart grids or renewable energies) despite political statements and small-scale experimentation.	•	Coupling IT and PEV industry by promoting and establishing partnerships and joint-ventures to realise development of shared, connected and autonomous vehicles [146,151].
7. Resource mobilisation and allocation	7.1 Financial support	•	See examples in cost alteration (S5.1), industry nurturing (S5.2), knowledge production (S5.3) and infrastructure preparation (S7.2). Providing public loans and grants for R&D to automakers [165].	•	Instruments from phase one (see S5.1–5.3 and S7.2) continued.
	7.2 Infrastructure preparation	•	Providing government subsidies for establishing public or private charging infrastructure [180].  Directing national power utilities to install charging stations [173].	•	Instruments from phase one (see S7.2) continued. Mandating the installation of charging infrastructure in town planning and building codes [180].
	7.3 Human resource fostering		Allocating R&D funds to industry and research institutes [140].  Establishing new research institutes and innovation parks [150].		Instruments from phase one (see S7.3) continued [183].
Targeted strategies for a	ccelerating decline of in	cumbent	technologies		
8. Reducing incumbent technology production and use	8.1 Technology control and restriction	•	Restricting the purchase/usage of ICE vehicles while incentivising the purchase and usage of NEVs via driving day bans and reduced allocation of licence plates to ICE vehicles [169].		Instruments from phase one (see S8.1) continued. Setting mandatory ZEV production targets to supress the volume of new ICE passenger vehicle sales [133,163].
	8.2 Support removal	$\bigcirc$	Use of this strategy not observed.		Halting the issuance of permits to proposed constructions of new ICE vehicle factories [11,146]. Abolishing diesel subsidies for public bus fleets [132].
	8.3 Structural reforms	$\bigcirc$	Use of this strategy not observed.	•	Promoting the market entry of foreign firms by weakening protection measures (see S5.4) to stimulate competition and technological learning for domestic firms [144,146].
	8.4 Actor changes		Weak use of this strategy. Political statements made about merging redundant vehicle makers and reorganising state-owned enterprises to boost NEV development and production. Yet no evidence of implementation.		Implementing mergers and changing leadership of several state-owned vehicle makers (e.g. FAW, Dongfeng and Changan) to accelerate PEV production, innovation and cost-cutting via economies of scale [174].

consumment such inless in wilet cities [141]. Circa 2015, where two months of a stand due

indicates use of a strategy is 'weak' or 'not observed'. indicates use of a strategy is 'strong'. See Section 3.2 for further explanation.

government vehicles in pilot cities [141]. Since 2015, phase two marks a shift towards mass-market adoption and the widespread rollout of charging infrastructure [142,143].

Though dominated by the state, the network driving China's transition to electric mobility also includes actors from industry, academia, and research institutes [144,145]. Incumbent government actors include political elites (e.g., President Xi Jinping, electric mobility advocates like Wan Gang), national agencies (e.g., the National Development and Reform Commission, the Ministry of Science and Technology, the Ministry of Industry and Information Technology), and sub-national authorities at provincial or city level (e.g., Beijing, Shanghai, Shenzhen) [146,147]. Sub-national agencies actively experiment with new policies, but their mandate is strongly dictated by the national government [148]. Industry comprises incumbent state-owned automakers (e.g., FAW, Dongfeng), newcomer private automakers and battery makers (e.g., BYD, Geely, CATL), and industry alliances and associations (e.g., EV100). While newcomers are gaining business and political power, incumbent state-owned enterprises have much more influence over policymaking, thanks to government links [145]. Finally, research institutes include R&D centres in elite government universities (e.g., Tsinghua University, Tongji University). Government dominates this

web of actors due to the one-party system and control over state-owned enterprises and industry associations [145,149] (int. 3,5). Accordingly, our analysis focuses on transition strategies formulated by national government and, to a lesser extent, sub-national governments.

Three motivations underpin China's transition strategy. The first is environmental, linked to desires to reduce air pollution and carbon emissions from increasing automobile usage in major cities (int. 4,6) [137,150]. Concerns over civil uprising against government policy due to chronic and severe air pollution strengthen this ambition [38]. The second motivation concerns industrial upgrading and the government's desire to boost the international competitiveness of the domestic autoindustry [11] (int. 4). With China's mastery of internal combustion engines (ICE) lagging behind foreign automakers [144], the lower complexity of building PEVs provides a precious opportunity for technological leapfrogging [147,151]. Electrification also serves ambitions to foster China's global dominance of connected and autonomous vehicle technologies [146]. The third motivation is energy security. This arises from the need to curb China's growing consumption of imported oil for transport and reduce economic vulnerability to price increases and supply disturbances [152] (int. 5).

Table 4
Japan: Summary of strategies used.

		1 Hase	1: Pre-development and exploration (1999–2014)	r nase .	2: Acceleration (2015–2020)
Category	Strategy	Use*	Instrument examples	Use*	Instrument examples
Overarching coordination	n strategies				
Knowledge     management	1.1 Knowledge dissemination	•	Establishing industry alliances for knowledge sharing and joint R&D (e.g. Fuel Cell Commercialization Conference of Japan) [194].  Diffusing knowledge from government supported	•	Instruments from phase one (see S1.1) continued [128,194].
2. Actor networks and collaboration	2.1 Inclusive governance	$\bigcirc$	innovation activities via public forums [193]. Weak use of this strategy. The principal governance framework for hydrogen mobility (Strategic Hydrogen and Fuel Cell Strategy Council) was dominated by large industry [158].	$\bigcirc$	Situation unchanged from phase one (see S2.1) [157,158].
	2.2 State–industry collaboration	•	Japan Hydrogen and Fuel Cell Project) to support the development and commercialisation of hydrogen mobility and refuelling infrastructure [131]. Producing joint roadmaps and identifying areas for collaboration via government-industry committees [156].	•	Producing joint roadmaps and identifying areas for collaboration via government-industry committees [154].  Establishing international hydrogen supply chains with public and private investment and collaboration [185].
	2.3 Industry alliances	•	Establishing alliances of oil/gas, engineering, electronics and automotive firms along with research institutions to accelerate knowledge sharing, innovation and fuel cell commercialisation (e.g. Research Association of Hydrogen Supply/Utilization Technology and the Fuel Cell Commercialization Conference of Japan) [156,194,195].	•	Instruments from phase one (see S2.3) continued. Mobilising oil/gas companies, automakers and investors to accelerate deployment of refuelling infrastructure via Japan H2 Mobility (JHyM) platform [187].
3. Guidance and planning	3.1 Planning and commitments	•	Publishing government-industry roadmaps to spur long- term commitments to developing and commercialising FCEVs and fix quantitative targets for deployment, performance and cost [155,156,158,196].	•	Instruments from phase one (see S3.1) continued [154]. Publishing the multi-decade vision of a 'hydrogen society' to motivate stakeholders and spur innovation and investments [20,153].
	3.2 Monitoring and disclosure	$\bigcirc$	Use of this strategy not observed.	•	Monitoring and annually reporting progress toward government targets for vehicle and refuelling infrastructure development, R&D agendas and cost performance targets [197].
Targeted strategies for ac	ccelerating creation and 4.1 Influencing public views	diffusion	Conducting public demonstrations of FCEVs and fuel cell buses to raise public awareness, especially at high profile international events [198].  Conducting public forums, trade shows and information provision to motivate and educate politicians, industry	•	Instruments from phase one (see S4.1) continued [154].
	4.2 Political lobbying	$\bigcirc$	and the public.  Weak use of this strategy. No dedicated coalition for political advocacy was observed.	$\bigcirc$	Situation unchanged from phase one (see S4.2).
5. Industry creation and technology production	5.1 Cost alteration	•	Providing public subsidies to reduce construction costs of hydrogen refuelling stations [199].	•	Instruments from phase one (see S5.1) continued [186]. Setting voluntary price ceiling for hydrogen fuel fo transport (1,000 yen/kg) [128]. Providing public subsidies and stamp duty waivers to reduce purchase price of FCEVs and fuel cell buses [128,189].
	5.2 Industry nurturing	•	Establishing university-industry knowledge clusters (e. g. Fukuoka Hydrogen Strategy and the Yamanashi Fuel Cell Valley). Providing government funding for R&D and	•	Instruments from phase one (see S5.2) continued.
	5.3 Knowledge production	•	demonstrations to support industry creation [193]. Supporting basic and applied research via R&D and demonstration subsidies to industry and research institutions [156]. Establishing industry-academic research centres and knowledge clusters (e.g. Fuel Cell Nanomaterials Centre at the University of Yamanashi) [200].	•	Instruments from phase one (see S5.3) continued for R&D and demonstration subsidies [186,189,201] and knowledge clusters (e.g. Yamanashi Fuel Cell Valley) [186].
	5.4 Industry protection	$\bigcirc$	Use of this strategy was not observed.	$\bigcirc$	Situation unchanged from phase one (see S5.4).
6. Market creation and technology diffusion	6.1 Niche creation and experimentation	•	Establishing demonstration or pilot projects to test and showcase hydrogen mobility (FCEVs, buses, trucks), hydrogen production and business models [155].		Instruments from phase one (see S6.1) continued [186,202].
	6.2 Demand creation	$\bigcirc$	Weak use of this strategy. Limited evidence of explicit measures like subsidies to stimulate vehicle demand.	•	Providing public subsidies and stamp duty waivers to reduce purchase price of FCEVs and fuel cell buses
				_	[128,189]. Instruments from phase one (see S6.3) continued

Table 4 (continued)

		Phase 1: Pre-development and exploration (1999–2014)			2: Acceleration (2015–2020)
Category	Strategy	Use*	Instrument examples	Use*	Instrument examples
	6.4 Sectoral coupling	•	Establishing international standards for key technologies and processes (i.e. refuelling) [155,156]. Promoting the simultaneous development and commercialisation of fuel cells for mobile and stationary applications (i.e. mobility and heat/power cogeneration) through roadmaps and industry innovation platforms [156,194].	•	Instruments from phase one (see S6.4) continued, notably by simultaneously providing subsidies and setting diffusion targets for mobility and cogeneration [154].  Promoting consumption of hydrogen in the transport and electricity sector to achieve economies of scale [128,154].
7. Resource mobilisation and allocation	7.1 Financial support		Providing public subsidies to reduce construction costs of hydrogen refuelling stations [199].		Instruments from phase one (see S7.1) continued [186]. Providing operating revenue to refuelling stations via the H2 Mobility platform [128].
	7.2 Infrastructure preparation	•	Establishing an industry partnership (HySUT) to prepare a national network of refuelling infrastructure by testing and developing technology ahead of vehicle deployment [156,199].	•	Fixing multi-year deployment targets for refuelling stations [128,154]. Establishing an industry alliance (Japan H2 Mobility) to mobilise public-private funding for construction and operation of refuelling stations and plan network development [128,186].
	7.3 Human resource fostering	•	Establishing research institutes and training programmes at prefectural level (e.g. the Hy-Life Project in Fukuoka) [186,189].	•	Instruments from phase one (see S7.3) continued.
Targeted strategies for ac	celerating decline of inc	cumbent	technologies		
8. Reducing incumbent technology production and use	8.1 Technology control and restriction	$\bigcirc$	Use of this strategy not observed.	$\bigcirc$	Weak use of this strategy. The government intention to ban sales of ICE vehicles by mid-2030 was only announced in December 2020 [203].
£	8.2 Support removal	$\bigcirc$	Use of this strategy not observed.	$\bigcirc$	Use of this strategy not observed.
	8.3 Structural reforms	Ŏ	Use of this strategy not observed.	Ŏ	Use of this strategy not observed.
	8.4 Actor changes	$\bigcirc$	Use of this strategy not observed.	$\bigcirc$	Use of this strategy not observed.

# 4.2. Hydrogen mobility in Japan

Japan, like China, actively supports the production and diffusion of hybrid, battery, and hydrogen drivetrains. We focus on FCEVs due to the heavy emphasis on hydrogen in national energy and transport policies [20,153,154]. Matching the Californian case, phase one is set to 1999–2014<sup>4</sup> to coincide with a period of intensive technological development, testing and market preparation [155,156]. Phase two is set to 2015–2020. Not only do efforts by government and industry to commercialise FCEVs accelerate significantly during this phase, 2015 marks the start of commercial FCEV sales and serial production (notably the Toyota *Mirai*) and the national rollout of refuelling infrastructure.

indicates use of a strategy is 'weak' or 'not observed'. indicates use of a strategy is 'strong'. See Section 3.2.

A distinct network of domestically-based incumbent actors is driving the transition [157,158]. Government actors at national level include the Ministry of Economy, Trade and Industry; the Prime Minister and his Cabinet; and multiple agencies like the New Energy and Industrial Technology Development Organisation and the Ministry of Environment. Sub-national governments such as Tokyo Metropolitan, Yokohama, and Aichi Prefecture are also highly active. Incumbent firms include vehicle manufacturers (Toyota, Honda), engineering or electronic firms (e.g. Mitsubishi Heavy Industry, Kawasaki Heavy Industry, Toshiba, Panasonic), oil and gas suppliers (e.g. Iwatani, ENEOS), and industry alliances (e.g. Japan H2 Mobility). Multiple universities and research institutes also participate through technology development and policy advisories. Our case focuses on strategies used by government agencies (mainly national but also prefectural) and incumbent industry.

Three sets of motivations propel Japan's transition towards hydrogen mobility. First is climate change and energy security. With the majority of Japan's energy needs imported as hydrocarbons, large-scale deployment of renewable energy has historically been hampered by

geographical, political, technological, and economic factors [31,159]. Thus, state actors are looking to hydrogen to spur the decarbonisation of transport, electricity generation, and industry while opening up new energy import pathways (int. 14,16,19). The second motivation is economic. Not only is the production and export of hydrogen and fuel-cell technologies perceived as an important growth market, Japanese firms and research institutes have gained decades of expertise under state support [160]. The third motivation concerns industrial development. State actors and incumbent automakers are concerned by the global shift towards battery vehicles. With fewer technological hurdles to hamper new entrants, the growing PEV market could destroy the international competitiveness of Japanese automakers, attained from mastering highefficiency hybrid or ICE vehicles (int. 14,20). Also, when produced at scale, PEVs will eliminate the need for part suppliers and derivative businesses underpinning the incumbent vehicle industry's vertically integrated business model [146]. Moreover, the automotive industry is a major contributor to employment, GDP, and exports. Thus, industry and government alike are looking to the complexity of fuel-cell technology to protect against encroachment by new entrants [156] and preserve the basic structure of existing part supply chains (int. 14,16,19).

#### 4.3. Hydrogen mobility in California

With state policy focused on supporting zero-emission drivetrains (i. e. both PEVs and FCEVs), our analysis focuses on the latter, including passenger vehicles, buses, and trucks. Since similar forces shape Japan's and California's hydrogen-mobility markets, phase one (1999–2014) and phase two (2015–2020) are set uniformly for each case.

The principal incumbents from state-level government agencies whose actions we examine consist of the California Energy Commission, the California Air Resources Board, and the Office of Governor. For incumbent industry, we focus on collective strategies employed by foreign and domestic firms headquartered out-of-state, such as

<sup>&</sup>lt;sup>4</sup> Per footnote 3, this phase goes from January 1999 to December 2014.

Table 5 California: Summary of strategies used.

Targeted strategies for accelerating creation and diffusion 4. 1 Influencing public wiews  1. Implementing vehicle demonstrations aimed at the public and politicians via the California Fuel Cell Partnership and California Fuel Cell Partnership and California Fuel Cell Partnership and California Fuel Cell Introducing fiele cell buses into public transif fleets [221,222].  5. Industry creation and technology production  5. Industry creation and technology and california Fuel Cell Partnership and California Fuel Cell Fuel Partnership and California Fuel Cell Fuel Partnership and California Fuel Cell Partnership Fuel Cell Part			Phase ?	1: Pre-development and exploration (1999–2014)	Phase	2: Acceleration (2015–2020)
1. Exonotology management managem	Category	Strategy	Use*	Instrument examples	Use*	Instrument examples
1. Lacowiedge management discustimation and place of the Cell Shamming and without any source of an emphrical process of the Cell Shamming and without any source of an emphrical discustion of collaboration and place of the Cell Shamming and and policy trends. Control to diffuse market and policy trends. Control to diffuse market and policy trends. Limited use of this strategy. Steering groups in the process of the control of the market and policy trends. Limited use of this strategy. Steering groups in the process of the control of the process of the proc	Overarching coordination	n strategies				
2. Actor networks and 2. J. Medistry collisionation government and collisionation government and industry collisional particularly (California Ford Coll Partnership) dominated leading puller-protein proteinship of collisional partnership (California Ford Coll Partnership) dominated partnership (California Ford Coll Partnership) dominated industry activities and accelerate development of hydrogen mobility market [101].  S. Guidance and 3.1 Ploruning and commitment commitment and commitment and development of hydrogen mobility market [101]. Setting aid-series and long-term targets for vehicle and relations of more development of hydrogen mobility market [101]. Setting aid-series and long-term targets for vehicle and relations of more development and evelopment of more development and evelopment of more development and evelopment and disclosure of disclosure protection.  4. 2. Pollutical mid-lying and legally binding government industry roadinaps and legally binding and integrated annual report to disclose progress towards government/funtury targets for vehicle and rectangly and report to disclose progress towards government/funtury targets for vehicle and rectangly and report to disclose progress towards government/funtury targets for vehicle and rectangly and politicism is vehicle and rectangly and politicism and the protection of the protection and rectangly production and rectangly government and politicism and the protection of the	1. Knowledge	1.1 Knowledge	•	and Fuel Cell Summit) and webinars as well as publishing reports (e.g. by California Fuel Cell	•	Instruments from phase one (see S1.1) continued.
Listabilishing public-private partnership (California Ped Cell Partnership to a laging government and industry activities and accelerate development of public private partnership (California Ped Cell Partnership to a laging government and industry activities and accelerate development of public private partnership (Listabilishing and public private private) and private private and accelerate control of public private private and accelerate control of public private privat			$\bigcirc$	Limited use of this strategy. Steering groups in the principal governance framework for hydrogen mobility (California Fuel Cell Partnership) dominated	$\bigcirc$	
2.3 Industry ellionese Carbonium industry alliances (e.g. California Pued Carbonium Carb		•	•	Establishing public–private partnership (California Fuel Cell Partnership) to align government and industry activities and accelerate development of	•	=
3. Guidance and planning and commitments continued commitments continued commitments of continued commitments and consistent margets for whiched and continued commitments and lengthy binding government policy (security or control) [219].  3.2 Monitoring and disclosure of the strategy in the control [219]. Weak use of this strategy in 2014.  4. Legitimisation of collections of collections with the Cultifornia Fuel Cell public and politicisms with the Cultifornia Fuel Cell public from the Cell p		2.3 Industry alliances		Establishing industry alliances (e.g. California Fuel Cell Partnership and the California Hydrogen Business Council) to align industry activities and accelerate	•	
phase two (see S3.2) began in 2014.  Targeted strategies for accelerating creation and diffusion 4. Legitimisation 5. Legitimisation 6. Le				Setting mid-term and long-term targets for vehicle and refuelling station deployment via government- industry roadmaps and legally binding government	•	Instruments from phase one (see S-3.1) continued.
4. Legitimisation views   Implementing vehicle demonstrations aimed at the public and politicians via the California Hydrogean Business Council, Introducing to level buses into public transit the California Hydrogean Business Council, Introducing to level buses into public transit in the California Hydrogean Business Council to carry out political advocacy for fiel cell mobility and hydrogean [217]. Establishing a dedicated industry alliance (California Hydrogean Business Council) to carry out political advocacy for fiel cell mobility and hydrogean [217]. Establishing a co-funding scheme to cover portion of construction costs for hydrogen refuelling stations and technology production   S. 2. Industry		•	$\bigcirc$	Weak use of this strategy. Instruments described in		progress towards government/industry targets for vehicle and refuelling infrastructure development and
5. Industry creation and technology production  5. 2 Industry murruring  5. 2 Industry murruring  5. 3. Rowledge production  5. 4 Industry production  6. Market creation and technology diffusion  6. Market creation and technology diffusion  6. A Removing market barriers  6. 2 Demand creation  6. 3 Removing market barriers  6. 4 Sectoral coupling  6. 5 Setolations and industry-government and collaboration recognised and experimental from phase one (see S6.4) continued. Expanding the Low Carbon Fuel Standard to also provide revenue for hydrogen refuelling stations based on capacity [568,205].  6. Instruments from phase one (see S5.1) continued. Expanding the Low Carbon Fuel Standard to also provide revenue for hydrogen refuelling stations based on capacity [568,205].  6. Instruments from phase one (see S5.2) continued. Expanding the Low Carbon Fuel Standard to also provide revenue for hydrogen refuelling stations to support the expansion of new and existing companies building agree-mentiston fuels including hydrogen general true fuel fuels and refuelling stations and industry government funding for vehicles and refuelling stations for making feet (e.g., government funding for vehicles and refuelling stations in university campuses and public vehicle fleets [22], 222].  6. 2 Demand creation  6. 4 Sectoral coupling  6. 4 Sectoral coupling  6. 4 Sectoral coupling  6. 4 Sectoral coupling  7. 1 Setting minimum ZEV procurement targets for public bus agencies [225].  8. 2 Providing subsidies to reduce purchase costs of zero emission buses.  8. 2 Instruments from phase one (see S6.2) continued [226], incentivising FCEV p	-	4.1 Influencing public	diffusion	Implementing vehicle demonstrations aimed at the public and politicians via the California Fuel Cell Partnership and California Hydrogen Business Council. Introducing fuel cell buses into public transit fleets	•	Instruments from phase one (see S4.1) continued.
and technology production    Size   S		4.2 Political lobbying	•	Establishing a dedicated industry alliance (California Hydrogen Business Council) to carry out political	•	[218], with a new alliance (California Hydrogen
5.2 Industry murning  5.3 Knowledge production  5.4 Industry protection  6. Market creation and technology diffusion  6.2 Demand creation  6.3 Removing market barriers  6.4 Sectoral coupling  6.4 Sectoral coupling  6.4 Sectoral coupling  6.5 Rowledge production  5.5 Industry protection  6.6 Market creation and technology diffusion  6.5 Removing market barriers  6.6 Removing market barriers  6.7 Removing market barriers  6.8 Removing market barriers  6.9 Removing market barriers  6.1 Rectoral coupling  6.4 Sectoral coupling  6.5 Rowledge production was not experiment funding for vehicles and refuelling stations from phase one (see S5.3) unchanged. Situation from phase one (see S5.4) unchanged.  5. Situation from phase one (see S5.4) unchanged.  5. Situation from phase one (see S5.4) unchanged.  5. Situation from phase one (see S5.4) unchanged.  6. Situation from phase one (see S5.4) unchanged.  7. Testing and demonstrating fuel cell trucks and refuelling stations in university campuses and public bus agencies [2231, 2222].  8. Setting minimum ZEV procurement targets for public bus agencies [2251].  8. Providing subsidies to reduce purchase costs of zero emission buses.  8. Identifying problematic policies (e.g. regulations, laws and formal rules etc.) and proposing modifications via publications and industry-government dialogues [227].  8. Developing common standards and protocols for vehicle refuelling, stations afety through industry-government collaboration [227, 2228].  8. Promoting the use of hydrogen and fuel cells in nontransport areas such as renewable energy, energy storage and ce-generation via industry alliances (California Hydrogen Business Council) and public  8. Instruments from phase one (see S5.2) continued.  9. Testing and demonstrating FCEVs, fuel cell buses and refuelling stations in commercial fleets [224].  1. Instruments from phase one (see S6.2) continued [224].  1. Instruments from phase one (see S6.4) continued.  1. Instruments from phase one (see S6.4) continued.  1. Instruments from ph	and technology	5.1 Cost alteration	•	construction costs for hydrogen refuelling stations [223]. Establishing a market instrument (Low Carbon Fuel Standard) to generate credit-based revenue for refuelling stations for sales of low-carbon fuels	•	Expanding the Low Carbon Fuel Standard to also provide revenue for hydrogen refuelling stations
5.3 Knowledge production support focused on applied rather than basic research (e.g., government funding for vehicles and refuelling station demonstrations) [2221.  5.4 Industry protection 6. Market creation and technology diffusion of and experimentation and experimentation 6.1 Niche creation and experimentation 6.2 Demand creation 6.2 Demand creation 6.3 Removing market barriers 6.3 Removing market barriers 6.4 Sectoral coupling 6.4 Sectoral coupling 6.5 Setting minimum ZEV procurement targets for public bus agencies [225]. Providing subsidies to reduce purchase costs of zero emission buses. 6.4 Sectoral coupling 6.5 Setting minimum ZEV procurement targets for public bus agencies [225]. Providing subsidies to reduce purchase costs of zero emission buses. 6.4 Sectoral coupling 6.5 Setting minimum ZEV procurement targets for public bus agencies [225]. Providing subsidies to reduce purchase costs of zero emission buses. 6.5 Removing market barriers 6.6 Semoving market barriers 6.7 Removing market barriers 6.8 Removing market barriers 6.9 Removing market barriers 6.1 Niche creation and technology diffusion via publications and industry-government dialogues [217]. Developing common standards and protocols for vehicle refuelling, hydrogen storage and refuelling station safety through industry-government collaboration via industry yalliances (California Hydrogen Business Council) and public		•	•	Establishing government grants to support the expansion of new and existing companies building		Instruments from phase one (see S5.2) continued.
5.4 Industry protection 6. Market creation and technology diffusion technology diffusion technology diffusion 6.1 Niche creation and experimentation  6.2 Demand creation  6.3 Removing market barriers  6.4 Sectoral coupling  6.4 Sectoral coupling  6.4 Sectoral coupling  6.5 Industry protection 6.6 Industry campuses and public vehicle fleets [221,222].  Setting minimum ZEV procurement targets for public bus agencies [225]. Providing subsidies to reduce purchase costs of zero emission buses.  Setting minimum ZEV procurement targets for public bus agencies [225]. Providing subsidies to reduce purchase costs of zero emission buses.  Identifying problematic policies (e.g. regulations, laws and formal rules etc.) and proposing modifications via publications and industry-government dialogues [217]. Developing common standards and protocols for vehicle refuelling, hydrogen storage and refuelling stations ain dustry-government collaboration [227,228].  Promoting the use of hydrogen and fuel cells in nontransport areas such as renewable energy, energy storage and co-generation via industry alliances (California Hydrogen Business Council) and public  Situation from phase one (see S5.4) unchanged.  Testing and demonstrating fuel cell trucks and refuelling stations in commercial fleets (e.g. Project Portal and Shore-to-Store project in Los Angeles ports)  [224].  Instruments from phase one (see S6.2) continued.  Situation from phase one (see S6.2) continued.  Testing and demonstrating fuel cell trucks and refuelling stations in commercial fleets (e.g. Project Portal and Shore-to-Store project in Los Angeles ports)  [224].  Instruments from phase one (see S6.2) continued.  Situation from effecting project in Los Angeles ports and refuelling stations in commercial fleets (e.g. Project Portal and Shore-to-Store project in Los Angeles ports)  [224].  Instruments from phase one (see S6.3)			$\bigcirc$	Weak use of this strategy. Knowledge production support focused on applied rather than basic research (e.g. government funding for vehicles and refuelling	$\bigcirc$	Situation from phase one (see S5.3) unchanged.
6. Market creation and technology diffusion  6.1 Niche creation and experimentation  6.2 Demand creation  6.2 Demand creation  6.3 Removing market barriers  6.3 Removing market barriers  6.4 Sectoral coupling  6.4 Sectoral coupling  7 Esting and demonstrating FCEVs, fuel cell buses and refuelling stations in university campuses and public vehicle fleets [221,222].  8 Esting minimum ZEV procurement targets for public bus agencies [225].  Providing subsidies to reduce purchase costs of zero emission buses.  1 Identifying problematic policies (e.g. regulations, laws and formal rules etc.) and proposing modifications via publications and industry-government dialogues [217].  Developing common standards and protocols for vehicle refuelling, hydrogen storage and refuelling station safety through industry-government collaboration [227,228].  Promoting the use of hydrogen and fuel cells in non-transport areas such as renewable energy, energy storage and co-generation via industry alliances (California Hydrogen Business Council) and public		-	$\bigcirc$		$\bigcirc$	Situation from phase one (see S5.4) unchanged.
Setting minimum ZEV procurement targets for public bus agencies [225]. Providing subsidies to reduce purchase costs of zero emission buses.  6.3 Removing market barriers  Identifying problematic policies (e.g. regulations, laws and formal rules etc.) and proposing modifications via publications and industry-government dialogues [217].  Developing common standards and protocols for vehicle refuelling, hydrogen storage and refuelling station safety through industry-government collaboration [227,228].  Promoting the use of hydrogen and fuel cells in nontransport areas such as renewable energy, energy storage and co-generation via industry alliances (California Hydrogen Business Council) and public  Instruments from phase one (see S6.2) continued. ZEV procurement targets tightened to 100% by 2030 via the Innovative Clean Transit regulation [226]. Incentivising FCEV purchases with free fuel provision, access to car pool lanes and state subsidies [68]. Instruments from phase one (see S6.3) continued [2291], with the development of standards and protocols expanding to heavy-duty trucks [230].  Instruments from phase one (see S6.4) continued.  Instruments from phase one (see S6.4) continued.		6.1 Niche creation	•	refuelling stations in university campuses and public	•	refuelling stations in commercial fleets (e.g. Project Portal and Shore-to-Store project in Los Angeles ports)
Identifying problematic policies (e.g. regulations, laws and formal rules etc.) and proposing modifications via publications and industry-government dialogues [217].  Developing common standards and protocols for vehicle refuelling, hydrogen storage and refuelling station safety through industry-government collaboration [227,228].  Promoting the use of hydrogen and fuel cells in nontransport areas such as renewable energy, energy storage and co-generation via industry alliances (California Hydrogen Business Council) and public  Instruments from phase one (see S6.3) continued [229], with the development of standards and protocols expanding to heavy-duty trucks [230].  Instruments from phase one (see S6.4) continued.		6.2 Demand creation	•	bus agencies [225] . Providing subsidies to reduce purchase costs of zero	•	Instruments from phase one (see S6.2) continued. ZEV procurement targets tightened to 100% by 2030 via the Innovative Clean Transit regulation [226]. Incentivising FCEV purchases with free fuel provision,
vehicle refuelling, hydrogen storage and refuelling station safety through industry-government collaboration [227,228].  6.4 Sectoral coupling  Promoting the use of hydrogen and fuel cells in non-transport areas such as renewable energy, energy storage and co-generation via industry alliances (California Hydrogen Business Council) and public		_	•	and formal rules etc.) and proposing modifications via publications and industry-government dialogues [217].	•	Instruments from phase one (see S6.3) continued [229], with the development of standards and
transport areas such as renewable energy, energy storage and co-generation via industry alliances (California Hydrogen Business Council) and public		6.4 Sectoral counting		vehicle refuelling, hydrogen storage and refuelling station safety through industry-government collaboration [227,228].		Instruments from phase one (see S6.4) continued
(continued on next page)		2330. at conputing		transport areas such as renewable energy, energy storage and co-generation via industry alliances	•	•

Table 5 (continued)

		Phase 1: Pre-development and exploration (1999–2014)			Phase 2: Acceleration (2015–2020)			
Category	Strategy	Use*	Instrument examples	Use*	Instrument examples			
7. Resource mobilisation and allocation	7.1 Financial support	•	symposiums (e.g. the California Hydrogen and Fuel Cell Summit) [217].  Providing government funding to reduce the construction costs for refuelling stations [231].  Establishing a market instrument (Low Carbon Fuel Standard) to generate credit-based revenue for refuelling stations for sales of low-carbon fuels including hydrogen [223].	•	Instruments from phase one (see S7.1) continued [232], with the Low Carbon Fuel Standard expanded to also provide revenue for hydrogen refuelling stations based on capacity [68,205]. Providing government funding to support fuel cell truck demonstrations.  Automakers (Toyota and Honda) directly acquiring			
	7.2 Infrastructure preparation	•	Setting targets for refuelling station deployment [219]. Supporting the planning of refuelling station	•	equity in refuelling station companies (e.g. First Element) [233]. Instruments from phase one (see S7.2) continued [68,205].			
	7.3 Human resource fostering	•	deployment via government agencies and government- industry collaboration (California Fuel Cell Partnership) [219,227]. Providing grants (via the California Energy Commission's Clean Transportation Program) to support ZEV-related workforce training and education in industry and colleges [234].	•	Instruments from phase one (see S7.3) continued [235].			
Targeted strategies for ac	celerating decline of inc	umbent						
8. Reducing incumbent technology production and use	8.1 Technology control and restriction		Setting mandatory ZEV production targets to supress the volume of new ICE passenger vehicle sales.  [162,209] Setting minimum ZEV purchase requirements for vehicle fleets (10% by 2015) in public agencies [207].	•	Instruments from phase one (see S8.1) continued [209]. ZEV production targets were widened to target heavy-duty trucks via the Advanced Clean Trucks Regulation [211] while ZEV procurement targets for government agencies were raised to 25% by 2020. Setting a mandatory phase-out schedule for non-zero emission buses serving public fleets via the Innovative			
	8.2 Support removal	$\bigcirc$	Use of this strategy not observed.	•	Clean Transit Regulation [236]. Reducing then abolishing subsidies for the purchase of low-emission internal combustion engines in trucks and buses and redirecting to zero-emission technologies [214].			
	8.3 Structural reforms	•	Setting maximum carbon intensity limits for gasoline via the Low Carbon Fuel Standard to incentivize low-carbon transportation, including hydrogen [237].	•	Instruments from phase one (see S8.3) continued.			
	8.4 Actor changes	$\bigcirc$	Use of this strategy not observed.	$\bigcirc$	Use of this strategy not observed.			

automakers (e.g. Toyota, Honda, Hyundai, Nikola, Mercedes, GM, Cummins), fuel suppliers (e.g. Shell, Air Liquide and recently Chevron), and industry consortiums such as the California Hydrogen Business Council. A particularly influential entity is the public–private California Fuel Cell Partnership [161]. This mobilises industry and government around the shared goal of growing the FCEV market through vehicle demonstrations, infrastructure investment, collaborative planning, and outreach.

indicates use of a strategy is 'weak' or 'not observed'. indicates use of a strategy is 'strong'. See Section 3.2.

Like in China, government actors are chiefly motivated by environmental factors; namely air pollution and climate change [162] (int. 27,29,38,44). Los Angeles has historically suffered from some of the worst air quality in the U.S. Being the primary cause of this pollution, transport is also California's principal source of GHG emissions (around 40%). Mobility electrification is thus essential to achieve the ambitious goal of reducing emissions 40% by 2030 from 1990 levels [40] and improving air quality. Unlike Japan or China, California lacks a strong economic rationale to promote hydrogen mobility since automakers currently manufacturing FCEVs are based in Asia. Instead, state actors have historically promoted hydrogen mobility due to advantages like long driving ranges, suitability for large vehicles, and short refuelling times (int. 43).

#### 5. Findings

We now apply the analytical framework to identify the main strategies used by the incumbent actor network in each jurisdiction to spur the transition towards electric mobility. To facilitate comparison, each

case follows the same structure. After summarising prominent areas of application and typifying approaches, we consider variations over the two phases, notable outcomes, and challenges. Summaries of strategies used appear in Tables 3–5. We also provide more detailed descriptions and supporting evidence in online supplementary material.

# 5.1. Strategies used in China's electric mobility transition

As Table 3 shows, the Chinese state is employing all possible means to pursue its transition to electric mobility. Indeed, in phase two, only three strategies remain underused: inclusive governance (S2.1), monitoring and disclosure (S3.2), and political lobbying (S4.2). Several strategies emerged in phase two, such as sectoral coupling (S6.4) and accelerating the decline of ICE vehicles (C8). The mix has thus evolved in volume, complexity, and stringency.

Beginning with overarching coordination strategies, planning and commitment (S3.1) plays a crucial role, given the top-down nature of the transition and the need to mobilise scores of cities and provinces around the pursuit of common goals. Throughout both phases, multiple national government agencies have consistently issued multi-year development plans and statements of policy commitments [132,133]. These signal at regular intervals, for sub-national governments and industry, the specific and evolving areas of national priority (e.g., preparing charging infrastructure, establishing standards, accelerating the development and commercialisation of next-generation batteries and autonomous driving) and principles to guide these (e.g., market-led approaches). Development plans also fix numerical targets and roadmap-like

milestones. Examples include vehicle or charging pile deployment targets (e.g., 1 million NEV sales by 2020 and 3 million by 2025 under the *Made in China 2025* plan set in 2015 [163]) and technological performance targets (e.g. reducing average power consumption in pure PEVs to 12 kWh/100 km from the current 15 kWh, outlined in the *New Energy Vehicle Industry Development Plan 2021–2035*) [164].

Moving to targeted strategies for accelerating creation and diffusion, those in the industry-creation and technology-production category (C5) play a key role. First, mirroring tactics used to nurture industries like solar PVs [148], government actors have consistently exploited cost alteration (S5.1) to reduce manufacturing and purchasing costs for PEVs. Government subsidies are widely discussed in literature [163,165]. Awarded after a vehicle purchase, subsidies artificially suppress the cost of production and sale - benefitting both manufacturers and consumers. A noteworthy feature of subsidy instruments in phase two is the increasing stipulation that qualifying automakers reach minimum performance specifications (e.g. mileage per charge). This approach, which mitigates cost barriers, also nurtures industry (S5.2) by incentivising faster technological learning [146]. Second, many industry-protection instruments (S5.4) are employed. Scholars often discuss 'local protectionism', whereby sub-national governments have preferentially allocated subsidies to local automakers [132,137,139]. Yet discriminatory tactics have also been used nationally to protect the domestic NEV market from foreign automakers with cheaper or superior technologies [144,146,152,166]. Notably, foreign firms were disqualified from receiving public subsidies via a so-called 'whitelist' of eligible vehicle and battery manufacturers. <sup>5</sup> Tariffs on imported vehicles (25% in phase one, 15% in phase two) and the joint-venture requirement have also protected domestic industry. The latter policy mandated that foreign automakers wishing to produce vehicles within China to avoid import tariffs must partner with a domestic firm and transfer key NEV technologies.

Strategies for market creation and technology diffusion (C6) also feature prominently. The national and sub-national governments have widely used public procurement instruments to stimulate vehicle demand (S6.2), which also serves industry-nurturing objectives (S5.2). Specifically, the national government mandated minimum ratios of ZEVs when procuring new vehicles in government fleets, especially buses and taxis [132]. Next, despite abundant top-down managerial approaches, strategies to foster experimentation with novel technology, business, or policy arrangements (S6.1) also feature strongly. Most notably, the pilot city programme [141,167] supports the trialling of unique approaches based on local needs, from which promising outcomes are reflected into national policy [139]. Also noteworthy is that sectoral coupling measures (S6.4) intensified in phase two. Beyond mere electrification per se, the ultimate objective for mobility in China is to become the dominating global market for shared, interconnected, autonomous vehicles [146,168]. To this end, the government has promoted links between IT firms (e.g., Alibaba, Baidu, Tencent) and stateowned automakers (e.g., FAW, Dongfen) to drive cross-sector synergies and innovation [151].

China's case demonstrates an expanding and increasingly stringent suite of strategies to suppress and eventually abolish the production and use of ICE vehicles. Technology control and restriction (S8.1) began in phase one with instruments to curb demand for ICE vehicles. In 2011, governments in Beijing and other cities started reducing the number of licence plates issued to new vehicles via lottery systems while increasing the share for NEV counterparts [169]. In parallel, restrictions were introduced to curb ICE vehicle usage on particular weekdays or during periods of heavy smog. With such demand-side measures continuing

into phase two, stricter instruments such as no-go zones for ICE vehicles are currently being developed [170]. Since phase two, instruments to phase out ICE vehicles also began targeting production. First, automakers wishing to expand production capacity could only receive government permits if the scale of NEV production over the previous two years exceeded the national industry average [146]. Second, since 2019, quotas oblige automakers to produce a minimum and rising share of NEVs annually (starting at 10% and rising 2% yearly) [133,163]. Support removal (S8.2) has also occurred. After 2016, national and subnational governments stopped issuing permits for construction of new ICE vehicle factories [11,146]. Additionally, fuel subsidies for diesel bus fleets were abolished to encourage replacement with battery alternatives [132]. Finally, structural reforms (S8.3) and actor changes (S8.4) took place. Beginning in 2019, reforms have weakened protection instruments such as the abovementioned joint-venture requirement. By allowing foreign vehicle and battery makers to establish independently owned factories [144], this aims to stimulate faster technological learning and cost reduction by exposing domestic automakers to overseas competition [146,164].

Some strategies have been neglected or underexploited. Scholars critique the absence of instruments to remove market barriers (S6.3), such as national standards for vehicle and charging infrastructure during phase one [140,171]. Monitoring and disclosure (S3.2) is also weak. Although the Chinese state collects data to measure progress towards key targets such as vehicle and infrastructure diffusion, results are not publicly shared in a consistent, periodic fashion [137,140,150,167]. Political lobbying (S4.2) via dedicated industry platforms was not observed. In explaining this, interviewees emphasised the power of the political apparatus and cooperative relationships between state and industry in NEV development (int. 4,5).

These transition strategies have generated impressive results. From just 30,000 sales in 2013, China's on-road fleet of PEVs reached 2.58 million in 2020, more than double the size of all other countries combined [129]. In cities such as Shenzhen, public bus and taxi fleets have transitioned fully to batteries. Demand creation (S6.2) and cost alteration (S5.1) measures have contributed substantially to this market growth [163]. The number of domestic PEV and battery makers has exploded due to government subsidies to automakers (S5.1) and knowledge production support (S5.3) [150]. Automakers like BYD now export buses to Western markets; NIO produces long-range, high-performance vehicles rivalling Tesla [146]; and the driving range of batteries from newcomer manufactures like CATL exceeds that of incumbent Korean makers.

Despite these achievements, multiple challenges have emerged [144]. The pursuit of aggressive targets for vehicle sales sacrificed quality for quantity. This resulted in battery-safety problems like spontaneous fires or explosions, and low-performance vehicles whereby subsidies paid per unit of output reduced the incentive for firms to invest in R&D and technological improvement [147] (int. 1,2,5). Subsidies also triggered overproduction and excessive entry of new firms with similar technologies and poor economic competitiveness [148,163]. Some manufacturers cheated by over-reporting sales when applying for public subsidies [139]. Also, although protectionist instruments (S5.4) such as the joint-venture requirement have allowed domestic automakers to flourish, many over-rely on technology transferred from foreign partners [172]. This situation explains several tactical changes in phase two. For example, structural reforms (S8.3) aim to weed out weaker automakers by opening the market to foreign firms (int. 5). Additionally, the shift to a credit-based market trading system (inspired by California) seeks to reduce automakers' dependence on per-unit-based subsidies [146].

<sup>&</sup>lt;sup>5</sup> Introduced in 2015 then abandoned in 2019, this listed battery makers whose products could be eligible for government NEV subsidies. Japanese and Korean makers were initially excluded. See Schwabe (2020) and Kennedy (2020).

<sup>&</sup>lt;sup>6</sup> Reform of the joint-venture requirement is reported to apply only to new factories; see Schwabe (2020). Since trade tariffs remain in place, we consider industry-protection measures (S5.4) to be still 'strong' in phase two.

#### 5.2. Strategies used in Japan's hydrogen mobility transition

Table 4 summarises strategies collectively used by state and industry actors in Japan. The relative absence of instruments to suppress production and use of ICE vehicles is noteworthy (C8). In the other seven categories (C1–7), diverse strategies are used consistently, changing little over the two phases. The only two additions observed are monitoring and disclosure (S3.2) and demand creation (S6.2).

Beginning with overarching coordination strategies (C13), most involve tight collaborations between government and incumbent industry (S1.2). Collaborative actions include government-industry roadmap creation and committees advising government about key technological developments and market trends [154,156]. Tight government-industry links are also visible in publicly funded research and demonstration platforms. These include the Japan Hydrogen and Fuel Cell Project (implemented over 2002–2010), established to support the development of hydrogen transport and refuelling infrastructure [131], and projects seeking to establish international supply chains importing mass-produced hydrogen [185,186]. State actors also coordinated the formation of industry alliances (S2.3) that link competing firms and otherwise isolated knowledge producers to enable joint R&D, demonstrations and knowledge exchange. The ongoing Japan H2 Mobility (JHyM) platform [187], for example, mobilises energy vendors, automakers, and investors to spearhead the national rollout of refuelling infrastructure [186]. Government actors also leverage industry links to diffuse knowledge (S1.1) through forums and consortium-driven

State actors use planning and commitments (S3.1) to muster longterm engagements and investments from key private, academic, and sub-national-government stakeholders. Notable instruments include national policy documents and roadmaps [153] that articulate priority areas for innovation agendas, policy, and investments. These also set numerical targets and milestones for technology deployment, costs, and technical performance [154]. For example, ambitious FCEV deployment targets set by the state in phase one (50,000 cumulative sales by 2010, 5 million by 2020) were renewed in phase two (40,000 sales by 2020, 800,000 by 2030). Targets were also set for refuelling station numbers (160 by 2020, 320 by 2025), power density of fuel cells, platinum usage, costs of key vehicle components (e.g. fuel tanks, fuel cell stacks) and the economics and volume of hydrogen production [20,128]. In this suite of targets, state policymakers stressed those for vehicle deployment due to their expected role in motivating industry to upscale production and investments (int. 20).

Among planning and commitment instruments (S3.1), visions play a prominent role in championing hydrogen and fuel cells, as they have for other state-supported technology-development programmes like solar PVs [97]. In phase two, government and industry actors jointly formulated an aspiration to attain a 'hydrogen society' [20]. This envisions using hydrogen not just in vehicles but broadly across heavy industry, chemical production, heat and power generation, and other transportation like boats, trains, ships, and aeroplanes [153,154,188]. Three distinct but interconnected phases leading beyond 2040 were fixed, each signalling the evolving priority areas for R&D, policy, and business development. Japan's vision of a hydrogen society helps to guide sustained, long-term commitments towards materialising these phases.

Regarding strategies for accelerating creation and diffusion, the state's guiding hand also features prominently in knowledge production (S5.3). Particularly in phase two, government actors tried to accelerate innovation by encouraging competing automakers (e.g. Toyota, Honda) to openly share the technological problems hampering mass production of FCEVs. Thematic R&D grants support basic and applied research to spur corresponding solutions [128]. As in China and California, Japan's incumbents have manipulated market conditions to alter the costs of using technology (S5.1). Notable strategies include government subsidies for refuelling stations and vehicle purchases and an industry-wide price ceiling to reduce hydrogen-refuelling costs for motorists.

Government and industry also collaboratively provide financial support (S7.1) through the H2 Mobility platform. This subsidises operational expenses until on-road FCEV numbers have increased to the point where refuelling stations attain financial self-sufficiency. In this moneyintensive approach, around one third of Japan's hydrogen-related budget of 70 billion yen<sup>7</sup> in 2020 was devoted to subsidies [189,190].

Three strategies for market creation and technology diffusion (C6) are heavily exploited in both phases. First, niche creation and experimentation (S6.1) is actively used through demonstration projects. Implemented mostly by incumbent firms [158], these have involved public testing and showcasing of passenger vehicle, bus, and truck prototypes [155,191], refuelling stations, and mass production of hydrogen from fossil-fuel [185] or renewable energy [186,192]. Government agencies support this experimentation through funding and providing collaboration opportunities to otherwise competing firms. Second, state and industry actors have consistently underlined the need to remove market barriers (6.3) during both phases (int. 14,15,21). This two-fold approach involves establishing common standards and protocols for industry while simultaneously removing or reforming regulations that hamper market investments in hydrogen mobility (e.g. hydrogen safety laws) [128,155,156]. Third, sectoral coupling (6.4) strategies are actively levered. This is visible in efforts to establish a common supply of mass-produced, low-carbon hydrogen [128,188] bridging the mobility and power-generation markets. Specifically, the gigantic fuel requirements of gas-fired power plants are anticipated to benefit transport by lowering fuel-production costs through economies of scale (int. 18,19).

Throughout both phases, three strategies remain weakly used. First, unlike in California, Japanese incumbent industry promoting hydrogen mobility has not established a dedicated platform for political lobbying (S4.2). Although lobbying might occur through existing industry organisations representing the entire automotive industry, according to one respondent (int. 19), the cooperative and intimate nature of state–industry relations reduces the need for a dedicated lobbying entity. Second, protectionist measures (S5.4) were not observed. The same respondent explained that the economic importance of exporting automobiles has reduced the political feasibility of discriminating against foreign firms entering Japan's market. Third, as mentioned, strategies for accelerating the decline of incumbent ICE vehicle technologies (C8) are absent. Explaining this gap, multiple interviews (int. 14,19,22) cited resistance from incumbent industry and the political fear of triggering economic stress in one of Japan's most important industries.

This minutely orchestrated and planning-intensive transition strategy, though ambitious and comprehensive, has produced only mediocre results. For example, state policymakers have not achieved vehicledeployment aspirations fixed in both phases [131,193]. While phase two targets aimed for 40,000 cumulative FCEV sales by the end of March 2021, less than 10% of this was achieved by late 2020 [128]. The strategy of preparing infrastructure (S7.2) ahead of vehicles has been more successful. Some 135 refuelling stations were in operation in late 2020 (the government target is 160 by end of March 2021). This is undoubtedly due to generous government subsidies (S5.1, S7.1) and the success of industry alliance measures (S2.3) - whereby H2 Mobility is mobilising joint investments and overseeing network development. Yet refuelling companies see miserable prospects for profitability due to the low fuel requirements of the nascent vehicle population [190]. As one station operator lamented: 'We are doing this as a societal contribution' rather than for profit (int. 21).

This situation is the direct result of a lower-than-anticipated vehicle supply. Only two automakers (Toyota and Honda) mass-produced

<sup>&</sup>lt;sup>7</sup> Equivalent to around US\$ 661 million in February 2021.

 $<sup>^8</sup>$  In December 2020, Japan announced ambitions to ban the sale of gasoline engines after the mid-2030s, but this was not considered a 'consistent' measure for phase two.

vehicles during phase two. With their production volumes limited [128], the state's efforts to motivate other domestic automakers previously testing prototypes to enter mass production are yet to bear fruit [156]. This is largely explainable by supply-side barriers. These include the expense and technological complexity of mass-producing FCEVs and establishing production lines and part supply networks in addition to preferences for battery or hybrid powertrains in clean vehicle portfolios (int. 16,19,22).

#### 5.3. Strategies used in California's hydrogen mobility transition

Table 5 summarises strategies used by state and market actors in California. The approach is comprehensive and largely consistent over both phases. Only four strategies were underused in phase two. Of these, measures for knowledge production (S5.3) were deemed weak due to negligible support for science and early-stage R&D. Notably, California exhibits an expanding and increasingly stringent suite of strategies to accelerate the decline of vehicles with ICE engines (C8).

Beginning with actor networks and collaboration (C2), formalised interactions between state and incumbent industry (S2.2) feature strongly. The California Fuel Cell Partnership is a key enabler here. Established in 1999, this public-private organisation unites incumbent automakers, oil/gas suppliers, and manufacturers with government agencies to align policymaking with market-development efforts while enabling joint activities like planning vehicle and infrastructure deployment and carrying out research and outreach [161]. Viewed from the guidance and planning perspective (S3.1), California's approach, like China's and Japan's, is both focused on long-term objectives and highly managerial. For instance, state and industry actors have collaboratively fixed numerous milestones (S3.1) for vehicle and infrastructure deployment during both phases [68]. In phase two, targets include achieving 1,000 refuelling stations and 1 million on-road FCEVs by 2030 [204,205]. These form part of a suite of higher-level ZEV diffusion targets set as legally binding executive orders from successive governors (e. g. 1.5 million on-road ZEVs by 2025, set in 2012; then 5 million by 2030, set in 2018). These signalling instruments provide common objectives for industry and government, offering long-term stability in a dynamic democracy where the fate of technologies is often affected by changing preferences of political leaders at state and federal levels. California is the most proactive of the three cases at disclosing progress towards industry-government targets and other policy objectives (S3.2), with the California Air Resources Board publishing this annually [205] in a consistent, comparison-friendly format. Its commitment to data disclosure and transparency allows stakeholders to monitor and adapt activities to changes in the hydrogen-mobility market.

For industry creation and technology production (C5), the state introduced multiple instruments to improve the cost competitiveness of hydrogen mobility (S5.1). These mainly target refuelling stations, serving the goal of accelerating infrastructure deployment (S7.2). State actors have reduced the financial burden for refuelling-station developers by subsidising up to half of construction costs [206] and supporting income generation through regulation-backed and incentivebased market mechanisms [68]. To this end, the Low Carbon Fuel Standard generates credit-based payments in accord with hydrogen fuel sales and station capacity [205]. Finally, as in the Asian cases, state intervention extends to support industry nurturing (S5.2) and human resource development (S7.3), through grants for training programmes and for the expansion of ZEV-manufacturing capabilities. Although California does not possess a local FCEV manufacturing industry, this support appears driven by expectations of attracting new industries and creating ZEV-related employment opportunities as a political tactic [207,208].

In the category of market creation and technology diffusion (C6), niche creation and experimentation (S6.1) features heavily in both phases. Activities have mainly involved collaborations between government agencies and incumbent firms to test and showcase the market

readiness of vehicles (including buses and trucks) and refuelling infrastructure. Demonstrations and experiments also seek to generate applied knowledge about technological performance, user behaviour, vehicle compatibility with infrastructure, and business feasibility [68]. Thus, public demonstrations help support knowledge production (S5.3) and efforts to influence public views (S4.1). This said, knowledge production activities are focused overwhelmingly on applied research and market creation rather than early-stage R&D or basic science, which is supported by federal funding (int. 40,45). In explaining the applied focus of fuel-cell truck demonstrations, one bureaucrat (int. 29) stressed ambitions to 'scale up these projects to really prove their capabilities to the eventual market. Because, again, what we're trying to do is move that market faster than it would already do [without state intervention]'. Since demonstrations of fuel-cell trucks in the logistics chains serving Los Angeles ports also involve decarbonisation of storage and heating/ cooling, niche experimentation is a key lever for promoting sectoral coupling (S6.4) between mobile and stationary applications of hvdrogen.

California's transition approach includes strategies to phase out ICE vehicles (C8). Two pioneering instruments were introduced by the state in phase one. First, the ZEV mandate was formulated as a technology-control and restriction measure (S8.1). Adopted in 1990 and activated in 2005, this regulation suppresses the production of ICE vehicles by imposing a minimum and rising quota for annual ZEV sales while incentivising efforts beyond this through credit-trading [162,209,210]. At the end of phase two, this regulation expanded to target heavy-duty truck manufacturers [211]. Second, structural reforms (S8.3) have reduced institutional support for ICE vehicles. One instrument, the Low Carbon Fuel Standard, destabilises the institutional environment supporting the gasoline-station industry. By setting a declining maximum carbon-intensity standard for transportation fuels, it stimulates the supply of alternative low-carbon fuels such as hydrogen while increasing the cost of hydrocarbons [212].

Efforts to phase out ICE vehicles increased in volume and intensity during phase two. From around 2018, support removal (S8.2) actions began as state actors dismantled subsidies that previously incentivised the purchase of low-emission ICE buses and trucks, redirecting these to ZEV technologies [213,214]. Explaining this, one policymaker highlighted the influence of the aforementioned ZEV-diffusion targets set by state governors (S3.1), in particular their 'scale' and 'compressed timeframe' (int. 35). In parallel, control and restriction measures (S8.1) were tightened. Not only are the minimum quotas for ZEV sales rising each year, the minimum share of ZEV purchases required for government vehicle fleets was also raised, and a phase-out of non-electric vehicles was mandated.

These strategies have produced mixed outcomes. With more than 7,000 FCEVs in circulation by mid-2020 [205], California boasts the world's largest on-road fleet. With no locally based FCEV manufacturing, vehicles have been coaxed to California from Asia through proactive demand creation (6.2) and supply-side measures suppressing the production of ICE vehicles (S8.1). Through its sustained demonstration programmes (S6.1), California has established a reputation as an attractive test bed for emerging fuel-cell technologies for domestic and overseas automakers (int. 40). As for limitations, the speed of vehicle and refuelling-station deployment is below historical expectations, with planning targets consistently missed (S3.1). Slow growth of the vehicle population can be explained firstly by the limited size of the refuelling network (currently around 43 stations in early 2020) and its inability to support a larger fleet of FCEVs, secondly by the limited production volumes of Asian automakers (see Section 5.2), and thirdly by the shift of electrification strategies towards batteries by domestic and overseas automakers previously developing FCEVs [68,205] (int. 35,38,39,41,43).

#### 6. Discussion

#### 6.1. Comparing strategies and phases

We now identify common trends across the three cases and two transition phases. Table 6 compiles the strategies employed, to enable a bird's-eye comparison.

One striking observation is that, aside from a few variations discussed below, most strategies are commonly employed in each case – despite highly diverse socio-political and cultural contexts. The main temporal trend is that strategies are *added* rather than removed during the acceleration phase.

Among overarching coordination strategies, commonly used are those reflecting top-down, managerial approaches and tight links between state and incumbent firms such as state–industry collaboration (S2.2) and planning and commitments (S3.1). The absence of measures to involve diverse actors in the principal governance structures for each mobility technology was expected, but in phase two, only China lacks explicit strategies to publicly disclose progress towards government targets. 9

Considering targeted strategies for accelerating technological decline (C8) reveals sharp variations. Absent in Japan, they are used extensively in California and China. If viewed temporally, a ratcheting approach becomes visible, with the volume and stringency of strategies and associated instruments increasing in phase two. In China, this occurred notably through addition of structural reforms (S8.3) and actor changes (S8.4) by the state to stimulate mergers and increased ZEV production while removing institutional support for ICE vehicles. In California, government regulations for technology control and restriction (S8.1) and structural reforms (S8.3) were tightened or expanded (e.g. from passenger vehicles to trucks). Additionally, phase two saw an explicit strategy introduced to remove public subsidies (S8.2) for ICE trucks and buses.

Two findings merit emphasis given their slight deviation from transition approaches accentuated in literature. First, although legitimisation and advocacy (C4) occur in each country, strategies by industry in Japan and China are limited to influencing public views (S4.1). Political lobbying (S4.2) through dedicated industry entities was not observed. Interviewees in China and Japan echoed a view that close state-industry cooperation reduces the imperative for lobbying. A respondent in Japan (int. 19) stated: 'Because the state and industry are promoting hydrogen energy together, I don't see how the current situation requires political lobbying'. Similar observations were made for China (int. 1,3). It is possible that lobbying occurs via subtle or invisible channels. But evidence from China and Japan along with accounts in literature suggest that in developmental states, incumbent industry's influence on policy likely occurs through 'tight direct links that have formed over decades based on politics, bureaucracy and business' [80] rather than dedicated lobbying per se.

The second noteworthy finding concerns industry protection (S5.4), used exclusively in China. Undoubtedly, imitating strategies in other countries such as import tariffs and discrimination against foreign firms would lack political feasibility. But their importance in nurturing the technological capabilities of Chinese firms suggests that different approaches to protecting new supply-side industries merit more attention from transitions scholars. This is especially apparent given the emphasis on 'shielding niches' in transitions literature [85] and the prevalent use of industry-protection tactics in developmental states [41,75,84].

#### 6.2. Comparing contextual factors and instruments

Overall, our analysis has revealed similar patterns in strategy use across cases. Yet distinctions can be teased out by considering the contextual conditions influencing each transition approach and specific instruments used (e.g. regulatory or incentive-based), as Table 7 shows.

We begin with contextual details. With their incumbent vehicle industries, the transition to electric mobility in China and Japan was triggered by industrial development ambitions and the need to maintain automakers' technological competitiveness in an increasingly carbonsensitive and electrified global market. Lacking a local automotive industry, California's transition strategy was triggered by environmental aspirations, and specifically the need to mitigate chronic air pollution while reducing the state's largest source of climate emissions. Thus, the urgent environmental imperative to tackle poor air quality and reduce associated impacts on public health is exclusive to China and California.

Several sharper distinctions emerge if we consider the degree to which specific policy instruments are used to implement acceleration strategies. Here we consider two: regulation and market incentives. Other instruments like planning and public investments are also relevant to sustainability transitions [10,21]. Since these are widely used across all cases, we disregard them here.

The most notable difference lies in the use of stringent regulation. In California and China, command-and-control instruments were the driving force behind strategies aimed at cost alteration (S5.1), demand creation (S6.2), and technology control and restriction (S8.1). Here, environmental regulations were tightened or added to influence supplyside and demand-side behaviour. Examples include minimum ZEVproduction quotas and procurement rules for public fleets mandating a minimum share of zero-emission powertrains and a phase-out of fossilfuel counterparts. Regulation in China, seeking to increase public demand for ZEVs while curbing the growing ICE vehicle stock, even extended to affect individual behaviour through driving bans and restrictions on licence-plate allocation. No comparable regulations exist in Japan. 10 Not only does demand creation (S6.2) rely solely on subsidy instruments, Japan's general approach to regulation involves removing legal barriers, as visible for instance in efforts to dismantle market obstacles affecting refuelling stations (S6.3).

Further distinct variations emerge when considering the use of market incentives linked to environmental regulation. Critically, instruments distributing financial rewards based on environmental performance – as opposed to uniformly distributing fixed-subsidy amounts or ratios – are widespread in China and California. Here, environmental regulations mandate conformance to minimum standards (e.g. ZEV production ratios or the carbon intensity of transport fuel), then incentivise performance beyond this via opportunities for credit trading. China and California also encourage efforts to raise vehicles' technical performance by allocating ZEV credits or subsidies to manufacturers in accord with driving range. Again, Japan's transition approach differs. Instead of regulatory whips, Japanese state actors seek to influence the behaviour of incumbent industry through incentives such as subsidies and information provision.

The use of particular policy instruments can be explained by reconsidering the contextual conditions and ambitions to accelerate the decline of ICE vehicles (C8). Regarding regulation, chronic air pollution in China and California has historically provided a rationale for strict environmental regulation to accelerate the electrification of road transport (int. 4,6,27,29,38,44). The grave consequences for public health have also justified introducing supply-side and demand-side strategies to weaken the market share of ICE vehicles and their supporting institutions (C8). In California, the absence of an incumbent automotive industry lowers the political barriers to introducing rigorous

<sup>&</sup>lt;sup>9</sup> Various industry trade associations (e.g. China EV100 and China Association of Automobile Manufacturers) along with private think tanks disparately release data on vehicle sales or charging pile installations. Yet we were unable to find an integrated and annually published report from a government affiliated organisation with comprehensive data permitting year-to-year comparisons on progress towards state targets or policy outcomes. See Pelkonen (2018).

<sup>&</sup>lt;sup>10</sup> In late 2020, the Japanese government announced the intention to ban ICE vehicle sales – too late to count as 'strong' usage in phase two. See footnote 8.

**Table 6**Comparison of strategy use across cases.

		Phase 1: Pre-development & exploration			Phase 2: Acceleration		
Category	Strategy	China	Japan	California	China	Japan	California
Overarching coordination strategies 1. Knowledge management	1.1 Knowledge dissemination			•			•
2. Actor networks and collaboration	2.2 Inclusive governance		$\bigcirc$	$\tilde{\bigcirc}$		$\bigcirc$	$\tilde{\bigcirc}$
	2.2 State-industry collaboration						Ŏ
	2.3 Industry alliances			Ŏ			Ŏ
3. Guidance and planning	3.1 Planning and commitment			Ŏ			Ŏ
	3.2 Monitoring and disclosure	$\bigcirc$	$\bigcirc$	$\tilde{\bigcirc}$	$\bigcirc$		Ŏ
4. Legitimisation and advocacy	4.1 Influencing public views			Ŏ			Ŏ
	4.2 Political lobbying	$\bigcirc$	$\bigcirc$	Ŏ	$\bigcirc$	$\bigcirc$	Ŏ
5. Industry creation and technology production	5.1 Cost alteration			Ŏ			Ŏ
	5.2 Industry nurturing			Ŏ			
	5.3 Knowledge production			$\tilde{\bigcirc}$			
	5.4 Industry protection			$\tilde{\bigcirc}$			$\widetilde{\bigcirc}$
6. Market creation and technology diffusion	6.1 Niche creation and experimentation						
	6.2 Demand creation		$\tilde{\bigcirc}$				
	6.3 Removing market barriers						
	6.4 Sectoral coupling	$\tilde{\bigcirc}$					
7. Resource mobilisation and allocation	7.1 Financial support						
	7.2 Infrastructure preparation						
	7.3 Human resource development						
8. Reducing incumbent technology production and use	8.1 Technology control and restriction		$\tilde{\bigcirc}$			$\tilde{\bigcirc}$	
	8.2 Support removal		$\widetilde{\bigcirc}$			$\widetilde{\bigcirc}$	
	8.3 Structural reforms	$\widetilde{\bigcirc}$	$\tilde{\bigcirc}$	$\check{\bullet}$		$\widetilde{\bigcirc}$	
	8.4 Actor changes	Ŏ	Ŏ	$\overline{\bigcirc}$		Ŏ	$\overline{\bigcirc}$

indicates use of a strategy is 'weak' or 'not observed'. indicates use of a strategy is 'strong'. See Section 3.2.

controls. Moreover, its attractive status as America's largest vehicle market gives policymakers formidable negotiating power over automakers. As one state official remarked: 'If they want to sell cars here, they have to figure out how to play [by our rules]' (int. 38).

Also heavily influencing each country's transition approach are broader and historical governance norms. Chinese policymakers' ability to apply an iron fist can be explained by the vertical, single-party nature of the political apparatus and control over sub-national governments, state-owned enterprises, even private firms (int. 3,5,6). Commenting on this unique power, one respondent (int. 1) stated: 'In other countries, these policies are not easily introduced. Can you imagine in Japan if they [the state] said, "You cannot drive your car on Thursday? Or you have to go through a lottery to get car plates?" 'Beyond automobiles, California has a long history of introducing aggressive environmental mandates to 'push the envelope and set the regulations just a little beyond what is currently available [on the market]', as one stakeholder observed (int. 27). Policymakers are thus skilled at dealing with industry resistance and, most importantly, are backed by environmentally progressive voters (int. 44). These enabling conditions are lacking in Japan, where voluntary industry frameworks and roadmaps rather than stringent regulation are the historical norm for environmental governance [159]. Japan's absence of stringent regulations and decline strategies can also be explained by resistance from the automotive and gasoline industries [238] and the political difficulty of disrupting the economically important business models of incumbent automakers specialised in producing high-efficiency gasoline and hybrid engines (int. 14,19,22).

#### 7. Conclusions

This paper aimed to deepen knowledge of the nature of top-down strategies used by networks of incumbent state and market actors to accelerate socio-technical transitions towards sustainability. Our comparison of strategies used in China, Japan, and California to spur the production and diffusion of battery and hydrogen mobility made two concrete contributions to scholarship. First, we married knowledge from sustainability transitions and the developmental state to identify mechanisms that drive socio-technical and industrial transformation, combining these into a comprehensive framework. This approach provides an important opportunity to overcome the limited ability of transitions scholarship to theorise the role of incumbents and top-down, managerial approaches. Second, in applying this framework to learn how such strategies work in three geographies underrepresented in scholarship, we built a rich repository of data on contemporary governance practices used in three countries at the forefront of electric mobility transitions, also identifying conditions affecting the choice of particular instruments.

What did our study reveal about the nature of top-down transitions? First, and not surprisingly, the main transition agents in each country are distinct communities of incumbents from government (both national and sub-national) and powerful firms such as automakers, fuel providers, and technology manufacturers. While elite knowledge producing institutions contribute to these transitions, civil society actors play a negligible role. Next, the incumbent actors use meticulously coordinated managerial strategies to guide markets, spur innovation, and influence supply-side and demand-side behaviour. Regarding the use of specific instruments, we observed active use of traditional policies (e.g.

**Table 7**Defining attributes of each mobility transition.

	China	Japan	California			
Contextual factors						
Agency of industry in governance	Limited	High	High			
Motivations (for state actors)	Environmental amelioration (air pollution) Climate change mitigation Energy security Development of domestic automotive industry	Climate change mitigation Energy security Development of domestic automotive industry	Environmental amelioration (air pollution) Climate change mitigation			
Enabling or limiting factors	Authoritative power of central and local government over industry and society Proliferation of state-owned enterprises under government control	Historical preference for industry-led voluntary frameworks in environmental governance Presence of incumbent ICE vehicle industry	Historical acceptance of state- led regulation in environmental governance Absence of incumbent ICE vehicle industry			
Transition obje	ctive					
Objective	Creation and destruction	Creation only	Creation and destruction			
Specific instrun Use of stringent regulation	nents Strong Used to influence supply-side and demand-side behaviour (S8.1, S8.3)	Weak Reformed to remove market barriers (S6.3)	Strong Used to influence supply-side and demand-side behaviour (S8.1)			
Use of market incentives	Strong Performance-based subsidies to automakers (S5.1) Credit-based ZEV production quotas (S8.1)	Weak Absence of performance-based incentives Reliance on uniform subsidy amounts/ ratios and public investment (S5.1, S6.2)	Strong Credit-based financial support to refuelling stations (S5.1) Credit-based ZEV production quotas (S8.1)			

regulation, planning, subsidies, public investment and information) emphasised by developmental-state and governance scholars. Yet cases also revealed use of instruments championed by transition scholars (e.g. network building, visioning, niche-experimentation, and market-based incentives). Thus, as stated elsewhere [10,18], effective top-down transitions may involve a mix of contrasting governance paradigms (e.g. command-and-control, market-driven, network-driven). Scholars exploring incumbent-led transitions in other settings should therefore heed overlaps and conceptual blurriness when applying category-based theories to the messy reality of real-world cases.

Despite different political and social circumstances, findings revealed much resemblance in the core strategies driving each country's electric-mobility transition. Beneath the surface of generic strategies, however, we observed strong variation in the specific instruments used.

These were particularly accentuated by comparing the use of regulation and performance-based market incentives, and ambitions to phase out ICE vehicles. As observed elsewhere [11], these variations are shaped by contextual factors, such as motivations in environmental amelioration and industrial development, incumbent industry's influence on policy, and the degree of regulatory authority wielded by state actors. We thus found that a politically feasible transitions approach in one country might not be in another.

By highlighting the strong agency exercised by incumbents in pursuing the transition to electric mobility, our study can enrich future scholarship. It reaffirms a stated need [16,125] to move past theoretical tendencies – whether implicit or explicit – to view newcomers and bottom-up approaches positively, while negatively portraying elites and policy-heavy, managerial strategies. Examining alternative governance approaches in regions thinly represented in literature might therefore deepen our understanding of the diverse tools available for accelerating socio-technical transitions. Along with other research [6,36,45,52], our study explicitly sought to cross-fertilise theories of socio-technical transitions with causal explanations of state-led industrial transformation. Yet the developmental state is just one of many fields with abundant knowledge on various mechanisms that state actors might use when pursuing socio-technical change, and conditions that can affect strategy formulation.

Finally, methodological limitations provide cues for future scholarship. First, when measuring the degree to which strategies were used in each case, we adopted broad definitions and a coarse binary evaluation that resulted in mostly 'strong' rather than 'weak/not observed' outcomes. Future studies could fix more stringent criteria or a wider spectrum of values [136] to generate finer-grained understanding of the relative importance of different strategies. Second, we did not systematically measure interactions between strategies [67,239]. Future work could tackle this by drawing on policy-mix literature and paying attention to synergistic and antagonistic relations [67,239,240]. Third, our choice to study ongoing, still-evolving cases has limitations. If full market penetration for each technology is taken as the desired end point, our cases provide insights into relevant strategies only during the market-formation phases of 'pre-development and exploration' and 'acceleration'. Finally, given the interlinked nature and spill-over effects of individual strategies, we refrained from speculating about their impacts. Future studies could focus more on this topic.

# **Declaration of Competing Interest**

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

#### Acknowledgements

The first author is indebted to the many organisations and individuals who kindly cooperated for interviews and provided data. Many thanks to Dr Scott Valentine and Dr Benjamin Sovacool for providing valuable insights when planning this manuscript. This research was generously supported by Kaken funds (grant numbers 21H03666, 19K20501 and 19H04333) from the Japan Society for the Promotion of Science.

Appendix A. Summary of interview details

California

Period conducted No. of interviews (respondents)\* Representative affiliations of respondents\*\*

(continued on next page)

# (continued)

Period conducted	No. of interviews (respondents)*	Representative affiliations of respondents**	
Mar – May 2020	20 (26)	Government: California Air Resources Board; California Energy Commission; California Governor's Office of Business & Economic Development; Port of Los Angeles Public-private: California Fuel Cell Partnership Private firms: Air Liquide USA; Honda (American Honda Motor Inc.); Hyundai-Kia (America Technical Center Inc.) Shell New Energies; True Zero (First Element Fuel Brand); Toyota (Toyota Motor North America) Universities: University of California, Berkley (Institute of Transportation Studies); University of California, Davis (Institute of Transportation Studies) Non-profit: Center for Transportation and the Environment; International Council on Clean Transportation	
China Nov 2019	6 (9)	Universities: Shenzhen University (College of Management); Tsinghua University, Shenzhen Graduate School (Research Center for Modern Logistics), University of California, Davis (China Center for Energy and Transportation)  Private firms: WeBank  Non-profit: Rock Environment and Energy Institute	
Japan Jan 2019 – Mar 2020	18 (26)	Government: Ministry of Trade, Economy and Industry; New Energy Development Organisation; Tokyo Metropolitan Government Private firms and research institutes: Honda and Honda Central Laboratories; Iwatani; Japan H2 Mobility; Japan Research Institute; Kawasaki Heavy Industries; Mizuho Research Institute; Toyota Central Laboratories; Toyota University: (International Institute for Carbon Neutral Research); Musashino University; Tama University	
Total	44 (61)		

Notes: \* Four interviews, included here, were conducted outside this period as follows: Japan (March 2018), China (March 2020 and January 2021) and California (February 2021). \*\* Not all affiliations are listed.

# Appendix B Indicators for evaluating strategy use in each country

Category	<b>S</b> trategy	Indicator*
Overarching coordination strategies		
1. Knowledge management	1.1 Knowledge dissemination	"Strong" if there are multiple, explicit and consistent** measures to foster knowledge exchange and dissemination across government and industry such as information sharing networks, symposiums, publications, collaborations etc.
2. Actor networks and collaboration	2.1 Inclusive governance	"Strong" if there are multiple, explicit and consistent measures to balance the participation of incumbent actors with diverse social actors (e.g. NGOs, citizen groups) in the principal governance framework or decision-making body.
	2.2 State–industry collaboration	"Strong" if there are multiple, explicit and consistent collaborations between government and incumbent industry such as public–private partnerships/projects, foundation of public–private organisations or production of joint outputs like roadmaps, etc.
	2.3 Industry alliances	"Strong" if there are multiple, explicit and consistent measures to foster the development of joint actions across industry such as partnerships, coalitions, foundations or joint projects/investments etc.
3. Guidance and planning	3.1 Planning and commitment 3.2 Monitoring and	"Strong" if explicit goals/targets for production/diffusion etc. or formalised commitments to support the technology and associated socio-technical arrangements and encourage investments have been issued. "Strong" if progress towards key targets for vehicle and infrastructure deployment are monitored and
	disclosure	disclosed annually to the public in an integrated format (e.g. report) allowing year-to-year comparisons.
Targeted strategies for accelerating cr	reation and diffusion	
4. Legitimisation and advocacy	4.1 Influencing public views	"Strong" if there are multiple, explicit and consistent measures to create a positive image or reputation of the new technology through marketing, public-relation campaigns (in conventional and social media), and showcase events etc.
	4.2 Political lobbying	"Strong" if there are multiple, explicit and consistent measures by firms to influence government policy or support through dedicated advocacy coalitions.
5. Industry creation and technology production	5.1 Cost alteration	"Strong" if there are multiple, explicit and consistent measures to assist firms to reduce the cost of producing, purchasing or using the targeted technology or to generate extra income or reward good behaviour (e.g. subsidies, pricing schemes, carbon pricing schemes etc.).
	5.2 Industry nurturing	"Strong" if there are multiple, explicit and consistent measures to support the creation or expansion of new or larger technology producers to increase the supply of technology (e.g. subsidies for R&D or production, tax credits, land provision, formation of technology parks or regional clusters, local content requirements, etc.).
	5.3 Knowledge production	"Strong" if there are multiple, explicit and consistent measures that support the creation of both basic and applied knowledge (i.e. both scientific and practical in nature) such as funding for R&D, technology parks, research centres and demonstration projects etc.
	5.4 Industry protection	"Strong" if there are multiple, explicit and consistent policies or institutions in place to protect domestic industry (especially technology manufacturers) from external/overseas competition such as discriminatory subsidies, rules/laws, import tariffs etc.
6. Market creation and technology diffusion	6.1 Niche creation and experimentation	"Strong" if there are multiple, explicit and consistent efforts and support mechanisms for experimentation and the testing of novel technologies, policies, business models and social arrangements such as funding schemes, public demonstrations, pilot projects, innovation districts etc.
	6.2 Demand creation	"Strong" if there are multiple, explicit and consistent measures that target technology users to increase or stimulate purchases for new technology such as economic incentives (e.g. purchase subsidies or tax waivers), non-economic incentives (e.g. priority use of car pool lanes) or public procurement initiatives.
	6.3 Removing market barriers	"Strong" if there are multiple, explicit and consistent measures to both (i) reform policies or formal institutions (e.g. standards and laws) and (ii) establish technical standards, in the goal of reducing market obstacles for new technologies or investments.
	6.4 Sectoral coupling	
		(continued on next page)

#### (continued)

Category	Strategy	Indicator*
		"Strong" if there are multiple, explicit and consistent measures to link the production or usage of electric mobility or related technologies across different markets to accelerate overall technological advancement, synergies or cost reduction via economies of scale etc.
7. Resource mobilisation and allocation	7.1 Financial support	"Strong" if there are multiple, explicit and consistent measures that directly supply financial capital or revenue to support the production, demonstration or diffusion of new technologies and associated infrastructure or business models such as subsidies, grants, loan programs, market incentives etc.
	7.2 Infrastructure preparation	"Strong" if there are multiple, explicit and consistent support schemes or investments by government or industry for refuelling or recharging stations such as funding schemes, policy targets, regulatory changes, industry platforms etc.
	7.3 Human resource development	"Strong" if there are explicit measures to accelerate the development of human resources such as training programs or financial or institutional support for industry and research/educational institutions.
Targeted strategies for accelerating d	ecline	
8. Reducing incumbent technology production and use	8.1 Technology control and restriction	"Strong" if there are explicit and mandatory government policies to restrict the production, sale or use of ICE vehicles such as environmental standards (e.g. emissions regulations) or control policies (e.g. driving or production bans, production quotas, etc.) in the goal of accelerating the transition to electric mobility.
	8.2 Support removal	"Strong" if there are explicit government policies that reduce or remove existing support or incentives for ICE vehicles such as reductions in subsidies, R&D schemes, institutional incentives (e.g. tax reductions) etc. in the goal of accelerating the transition to electric mobility.
	8.3 Structural reforms	"Strong" if there are explicit policies that alter the wider market or regulatory environment to promote new entrants, accelerate production/adoption of ZEVs or weaken the institutional arrangements supporting the incumbent ICE vehicle industry (e.g. carbon standards or increased taxes for the production/sale of fossilfuel for transport or the sale/use of ICE vehicles, etc.).
	8.4 Actor changes	"Strong" if there are explicit reforms (e.g. abolishment, changes, mergers etc.) made in the organisational structure of government or industry decision-making bodies or ministries/agencies to increase the representation or influence of new actors in the goal of accelerating the transition to electric mobility and reducing the production or usage of ICE vehicles.

#### Notes:

#### Appendix C. Supplementary data

Supplementary data to this article can be found online at https://doi.org/10.1016/j.erss.2021.102184.

#### References

- B. Turnheim, B.K. Sovacool, Forever stuck in old ways? Pluralising incumbencies in sustainability transitions, Environ. Innovat. Soc. Trans. 35 (2020) 180–184, https://doi.org/10.1016/j.eist.2019.10.012.
- [2] B. Ghosh, J. Schot, Towards a novel regime change framework: Studying mobility transitions in public transport regimes in an Indian megacity, Energy Res. Social Sci. 51 (2019) 82–95, https://doi.org/10.1016/j.erss.2018.12.001.
- [3] A. Bergek, C. Berggren, T. Magnusson, M. Hobday, Technological discontinuities and the challenge for incumbent firms: destruction, disruption or creative accumulation? Res. Policy 42 (6) (2013) 1210–1224, https://doi.org/10.1016/j. respol.2013.02.009.
- [4] A. van Mossel, F.J. van Rijnsoever, M.P. Hekkert, Navigators through the storm: a review of organization theories and the behavior of incumbent firms during transitions, Environ. Innovat. Soc. Trans. 26 (2018) 44–63, https://doi.org/ 10.1016/j.eist.2017.07.001.
- [5] R. Bohnsack, A. Kolk, J. Pinkse, C.M. Bidmon, Driving the electric bandwagon: the dynamics of incumbents' sustainable innovation, Bus. Strategy Environ. 29 (2) (2020) 727–743, https://doi.org/10.1002/bse.2430.
- [6] P. Johnstone, P. Newell, Sustainability transitions and the state, Environ. Innovat. Soc. Trans. 27 (2018) 72–82, https://doi.org/10.1016/j.eist.2017.10.006.
- [7] P. Newell, A. Simms, How did we do that? Histories and political economies of rapid and just transitions, New Polit. Econ. (2020) 1–16, https://doi.org/ 10.1080/13563467.2020.1810216.
- [8] Roberts, C. and F.W. Geels, Political acceleration of sociotechnical transitions: Lessons from four historical case studies, in Transitions in Energy Efficiency and Demand: The Emergence, Diffusion and Impact of Low-Carbon Innovation, K. Jenkins and D. Hopkins, Editors. 2019, Routledge, Earthscan: London and New York. p. 177-194.
- [9] C. Roberts, F.W. Geels, Conditions for politically accelerated transitions: Historical institutionalism, the multi-level perspective, and two historical case studies in transport and agriculture, Technol. Forecast. Soc. Chang. 140 (2019) 221–240, https://doi.org/10.1016/j.techfore.2018.11.019.
- [10] C. Roberts, F.W. Geels, Conditions and intervention strategies for the deliberate acceleration of socio-technical transitions: lessons from a comparative multi-level analysis of two historical case studies in Dutch and Danish heating, Technol. Anal. Strateg. Manage. 31 (9) (2019) 1081–1103, https://doi.org/10.1080/ 09537325.2019.1584286.

- [11] J. Meckling, J. Nahm, The politics of technology bans: industrial policy competition and green goals for the auto industry, Energy Policy 126 (2019) 470–479, https://doi.org/10.1016/j.enpol.2018.11.031.
- [12] D. Rosenbloom, A. Rinscheid, Deliberate decline: An emerging frontier for the study and practice of decarbonization. WIREs, Clim. Change e669 (2020), https://doi.org/10.1002/wcc.669.
- [13] C. Roberts, F.W. Geels, M. Lockwood, P. Newell, H. Schmitz, B. Turnheim, A. Jordan, The politics of accelerating low-carbon transitions: towards a new research agenda, Energy Res. Social Sci. 44 (2018) 304–311, https://doi.org/ 10.1016/j.erss.2018.06.001.
- [14] A. Cherp, V. Vinichenko, J. Jewell, M. Suzuki, M. Antal, Comparing electricity transitions: A historical analysis of nuclear, wind and solar power in Germany and Japan, Energy Policy 101 (2017) 612–628, https://doi.org/10.1016/j. enpol.2016.10.044.
- [15] J. Köhler, F.W. Geels, F. Kern, J. Markard, E. Onsongo, A. Wieczorek, F. Alkemade, F. Avelino, A. Bergek, F. Boons, L. Fünfschilling, D. Hess, G. Holtz, S. Hyysalo, K. Jenkins, P. Kivimaa, M. Martiskainen, A. McMeekin, M. S. Mühlemeier, B. Nykvist, B. Pel, R. Raven, H. Rohracher, B. Sandén, J. Schot, B. Sovacool, B. Turnheim, D. Welch, P. Wells, An agenda for sustainability transitions research: State of the art and future directions, Environ. Innovat. Soc. Trans. 31 (2019) 1–32, https://doi.org/10.1016/j.eist.2019.01.004.
- [16] B. Turnheim, F.W. Geels, Incumbent actors, guided search paths, and landmark projects in infra-system transitions: Re-thinking Strategic Niche Management with a case study of French tramway diffusion (1971–2016), Res. Policy 48 (6) (2019) 1412–1428, https://doi.org/10.1016/j.respol.2019.02.002.
- [17] Valentine, S., Modifying Recipes: Insights on Japanese Electricity Sector Reform and Lessons for China, in Asia after the Developmental State: Disembedding Autonomy, T. Carroll and D.S.L. Jarvis, Editors. 2017, Cambridge University Press: United Kingdom. p. 429-456.
- [18] F.W. Geels, S. Sareen, A. Hook, B.K. Sovacool, Navigating implementation dilemmas in technology-forcing policies: a comparative analysis of accelerated smart meter diffusion in the Netherlands, UK, Norway, and Portugal (2000–2019), Res. Policy 50 (7) (2021), 104272, https://doi.org/10.1016/j. respol 2021 104272
- [19] O. Ejderyan, F. Ruef, M. Stauffacher, Entanglement of top-down and bottom-up: sociotechnical innovation pathways of geothermal energy in Switzerland, J. Environ. Dev. 29 (1) (2019) 99–122, https://doi.org/10.1177/ 1070496519886008.
- [20] G. Trencher, J. Van der heijden, Contradictory but also complementary: National and local imaginaries in Japan and Fukushima around transitions to hydrogen

<sup>\*</sup> A <u>value</u> of 'weak or not observed' was allocated when the degree of utilisation was inferior to descriptions in the listed indicator or if no corresponding strategies were observed.

<sup>\*\*</sup> We take 'consistent' to mean the continuation of the same measure over multiple years (at least two) consecutively.

- and renewables, Energy Res. Soc. Sci. 49 (2019) 209–218, https://doi.org/10.1016/j.erss.2018.10.019.
- [21] J. Meadowcroft, What about the politics? Sustainable development, transition management, and long term energy transitions, Policy Sci. 42 (4) (2009) 323, https://doi.org/10.1007/s11077-009-9097-z.
- [22] F.W. Geels, Technological transitions as evolutionary reconfiguration processes: a multi-level perspective and a case-study, Res. Policy 31 (8) (2002) 1257–1274, https://doi.org/10.1016/S0048-7333(02)00062-8.
- [23] F.W. Geels, Disruption and low-carbon system transformation: progress and new challenges in socio-technical transitions research and the Multi-Level Perspective, Energy Res. Soc. Sci. 37 (2018) 224–231, https://doi.org/10.1016/j. erss.2017.10.010.
- [24] J. Rotmans, R. Kemp, M. van Asselt, More evolution than revolution: transition management in public policy, Foresight 3 (1) (2001) 15–31, https://doi.org/ 10.1108/14636680110803003
- [25] S. Sarkar, M. Pansera, Sustainability-driven innovation at the bottom: Insights from grassroots ecopreneurs, Technol. Forecast. Soc. Change 114 (2017) 327–338, https://doi.org/10.1016/j.techfore.2016.08.029.
- [26] C. Berggren, T. Magnusson, D. Sushandoyo, Transition pathways revisited: Established firms as multi-level actors in the heavy vehicle industry, Res. Policy 44 (5) (2015) 1017–1028, https://doi.org/10.1016/j.respol.2014.11.009.
- [27] P. Kivimaa, S. Laakso, A. Lonkila, M. Kaljonen, Moving beyond disruptive innovation: A review of disruption in sustainability transitions, Environ. Innovat. Soc. Trans. 38 (2021) 110–126, https://doi.org/10.1016/j.eist.2020.12.001.
- [28] A. Mori, How do incumbent companies' heterogeneous responses affect sustainability transitions? Insights from China's major incumbent power generators, Environ. Innovat. Soc. Trans. 39 (2021) 55–72, https://doi.org/ 10.1016/j.eist.2021.02.003.
- [29] K.C. Seto, S.J. Davis, R.B. Mitchell, E.C. Stokes, G. Unruh, D. Ürge-Vorsatz, Carbon lock-in: types, causes, and policy implications, Annu. Rev. Environ. Resour. 41 (1) (2016) 425–452, https://doi.org/10.1146/annurev-environ-110615-085934
- [30] G.C. Unruh, Understanding carbon lock-in, Energy Policy 28 (12) (2000) 817–830, https://doi.org/10.1016/S0301-4215(00)00070-7.
- [31] G. Trencher, A. Rinscheid, M. Duygan, N. Truong, J. Asuka, Revisiting carbon lock-in in energy systems: explaining the perpetuation of coal power in Japan, Energy Res. Soc. Sci. 69 (2020), 101770, https://doi.org/10.1016/j. erss.2020.101770.
- [32] J.H. Wesseling, A. Van der Vooren, Lock-in of mature innovation systems: the transformation toward clean concrete in the Netherlands, J. Cleaner Prod. 155 (2017) 114–124, https://doi.org/10.1016/j.jclepro.2016.08.115.
- [33] F.W. Geels, From sectoral systems of innovation to socio-technical systems: Insights about dynamics and change from sociology and institutional theory, Res. Policy 33 (6) (2004) 897–920, https://doi.org/10.1016/j.respol.2004.01.015.
- [34] M.M. Smink, M.P. Hekkert, S.O. Negro, Keeping sustainable innovation on a leash? Exploring incumbents' institutional strategies, Bus. Strategy Environ. 24 (2) (2015) 86–101, https://doi.org/10.1002/bse.1808.
- [35] G. Trencher, N. Healy, K. Hasegawa, J. Asuka, Discursive resistance to phasing out coal-fired electricity: narratives in Japan's coal regime, Energy Policy 132 (2019) 782–796. https://doi.org/10.1016/j.enpol.2019.06.020
- [36] Mazzucato, M. The Green Entrepreneurial State. 2015 [cited 2020 15 September]; Available from: https://www.sussex.ac.uk/webteam/gateway/file.php?name=2 015-28-swps-mazzucato.pdf&site=25.
- [37] Mah, D.N.-y., Y.-Y. Wu, J.C.-m. Ip, and P.R. Hills, The role of the state in sustainable energy transitions: A case study of large smart grid demonstration projects in Japan. Energy Policy, 2013. 63: p. 726-737 DOI: https://doi.org/ 10.1016/j.enpol.2013.07.106.
- [38] G.C. Chen, C. Lees, The New, Green, Urbanization in China: between authoritarian environmentalism and decentralization, Chinese Political Science Review 3 (2) (2018) 212–231, https://doi.org/10.1007/s41111-018-0095-1.
- [39] A. Kronsell, J. Khan, R. Hildingsson, Actor relations in climate policymaking: Governing decarbonisation in a corporatist green state, Environ. Policy Govern. 29 (6) (2019) 399–408, https://doi.org/10.1002/eet.1867.
- [40] D. Sperling, M. Nichols, California's pioneering transportation strategy, Issues Sci. Technol. 28 (2) (2012) 59–66.
- [41] Z. Öniş, The logic of the developmental state, Compar. Polit. 24 (1) (1991) 109–126, https://doi.org/10.2307/422204.
- [42] Chen, G.C., Governing Sustainable Energies in China. 2016: Palgrave Macmillan.
- [43] Woo-Cummings, M., ed. The developmental state. 1999, Cornell University Press: Ithaca, New York State.
- [44] Johnson, C., The Developmental State: Odyssey of a Concept, in The Developmental State, M. Woo-Cummings, Editor. 1999, Cornell University Press: Cornell. p. 32–60.
- [45] J. Meckling, The developmental state in global regulation: economic change and climate policy, Eur. J. Int. Relat. 24 (1) (2018) 58–81, https://doi.org/10.1177/ 1006.11770066
- [46] N. Andrews, C. Nwapi, Bringing the state back in again? The emerging developmental state in Africa's energy sector, Energy Res. Soc. Sci. 41 (2018) 48–58, https://doi.org/10.1016/j.erss.2018.04.004.
- [47] Castells, M., Four Asian Tigers with a Dragon Head: A Comparative Analysis of the State, Economy and Society in the Asian Pacific Rim, in States and Development in the Asian Pacific Rim, R. Appelbaum and J. Henderson, Editors. 1992, Sage: London. p. 33-70.
- [48] Wade, R., East Asia's Economic Success: Conflicting Perspectives, Partial Insights, Shaky Evidence. World Politics, 1992. 44(2): p. 270-320 DOI: 10.2307/2010449.

- [49] C.M. Dent, East Asia's new developmentalism: state capacity, climate change and low-carbon development, Third World Quart. 39 (6) (2018) 1191–1210, https:// doi.org/10.1080/01436597.2017.1388740.
- [50] G.C. Chen, C. Lees, Growing China's renewables sector: a developmental state approach, New Polit. Econ. 21 (6) (2016) 574–586, https://doi.org/10.1080/ 13563467.2016.1183113.
- [51] J. Meckling, J. Nahm, When do states disrupt industries? Electric cars and the politics of innovation, Rev. Int. Polit. Econ. 25 (4) (2018) 505–529, https://doi. org/10.1080/09692290.2018.1434810.
- [52] M. Swilling, J. Musango, J. Wakeford, Developmental states and sustainability transitions: prospects of a just transition in South Africa, J. Environ. Plann. Policy Manage. 18 (5) (2016) 650–672, https://doi.org/10.1080/ 1523098/2015.1107716.
- [53] D. Vazquez-Brust, A.M. Smith, J. Sarkis, Managing the transition to critical green growth: The 'Green Growth State', Futures 64 (2014) 38–50, https://doi.org/ 10.1016/j.futures.2014.10.005.
- [54] K.-H. Chien, Pacing for renewable energy development: the developmental state in taiwan's offshore wind power, Annals Am. Assoc. Geograph. 110 (3) (2020) 793–807, https://doi.org/10.1080/24694452.2019.1630246.
- [55] J. Markard, F.W. Geels, R. Raven, Challenges in the acceleration of sustainability transitions, Environ. Res. Lett. 15 (8) (2020), 081001, https://doi.org/10.1088/ 1748-9326/ab9468.
- [56] M.P. Hekkert, R.A.A. Suurs, S.O. Negro, S. Kuhlmann, R.E.H.M. Smits, Functions of innovation systems: a new approach for analysing technological change, Technol. Forecast. Soc. Change. 74 (4) (2007) 413–432, https://doi.org/ 10.1016/j.techfore.2006.03.002.
- [57] P. Kivimaa, F. Kern, Creative destruction or mere niche support? Innovation policy mixes for sustainability transitions, Res. Policy 45 (1) (2016) 205–217, https://doi.org/10.1016/j.respol.2015.09.008.
- [58] L. Kanger, B.K. Sovacool, M. Noorkõiv, Six policy intervention points for sustainability transitions: a conceptual framework and a systematic literature review, Res. Policy 49 (7) (2020), 104072, https://doi.org/10.1016/j. respol.2020.104072.
- [59] L. Frank, K. Jacob, R. Quitzow, Transforming or tinkering at the margins? Assessing policy strategies for heating decarbonisation in Germany and the United Kingdom, Energy Res. Social Sci. 67 (2020), 101513, https://doi.org/ 10.1016/i.erss.2020.101513.
- [60] J. Markard, The next phase of the energy transition and its implications for research and policy, Nat. Energy 3 (8) (2018) 628–633, https://doi.org/10.1038/ s41560-018-0171-7.
- [61] D. Rosenbloom, J. Markard, F.W. Geels, L. Fuenfschilling, Opinion: Why carbon pricing is not sufficient to mitigate climate change—and how "sustainability transition policy" can help, Proc. Natl. Acad. Sci. 117 (16) (2020) 8664–8668, https://doi.org/10.1073/pnas.2004093117.
- [62] J. Markard, R. Raven, B. Truffer, Sustainability transitions: an emerging field of research and its prospects, Res. Policy 41 (6) (2012) 955–967, https://doi.org/ 10.1016/j.respol.2012.02.013.
- [63] Geels, F.W., B. Turnheim, M. Asquith, F. Kern, and P. Kivimaa, Sustainability transitions: policy and practice. 2019: European Environment Agency.
   [64] Victor, D.G., F.W. Geels, and S. Sharpe. Accelerating the low-carbon transition:
- [64] Victor, D.G., F.W. Geels, and S. Sharpe. Accelerating the low-carbon transition: The case for stronger, more targeted and coordinated international action. 2019 [cited 2020 October 10]; Available from: https://www.energy-transitions.org/ publications/accelerating-the-low-carbon-transition/
- publications/accelerating-the-low-carbon-transition/.
  [65] A. Smith, A. Stirling, F. Berkhout, The governance of sustainable socio-technical transitions, Res. Policy 34 (10) (2005) 1491–1510, https://doi.org/10.1016/j.respol 2005 07 005
- [66] T. Meelen, J. Farla, Towards an integrated framework for analysing sustainable innovation policy, Technol. Anal. Strategic Manage. 25 (8) (2013) 957–970, https://doi.org/10.1080/09537325.2013.823146.
- [67] M. Dijk, E. Iversen, A. Klitkou, R. Kemp, S. Bolwig, M. Borup, P. Møllgaard, Forks in the Road to E-Mobility: an evaluation of instrument interaction in national policy mixes in northwest Europe, Energies 13 (2020) 475, https://doi.org/ 10.3390/en13020475.
- [68] G. Trencher, Accelerating the production and diffusion of fuel cell vehicles: Experiences from California, Energy Rep. 6 (2020) 2503–2519, https://doi.org/ 10.1016/j.egyr.2020.09.008.
- [69] Steinmueller, W.E., Economics of Technology Policy, in Handbook of the Economics of Innovation, B.H. Hall and N. Rosenberg, Editors. 2010, North-Holland. p. 1181-1218.
- [70] B. Turnheim, F.W. Geels, Regime destabilisation as the flipside of energy transitions: Lessons from the history of the British coal industry (1913–1997), Energy Policy 50 (2012) 35–49, https://doi.org/10.1016/j.enpol.2012.04.060.
- [71] A. Bergek, S. Jacobsson, B.A. Sandén, 'Legitimation' and 'development of positive externalities': two key processes in the formation phase of technological innovation systems, Technol. Anal. Strategic Manage. 20 (5) (2008) 575–592, https://doi.org/10.1080/09537320802292768.
- [72] F.W. Geels, A. McMeekin, J. Mylan, D. Southerton, A critical appraisal of sustainable consumption and production research: the reformist, revolutionary and reconfiguration positions, Global Environ. Change 34 (2015) 1–12, https:// doi.org/10.1016/j.gloenvcha.2015.04.013.
- [73] R. Kemp, J. Schot, R. Hoogma, Regime shifts to sustainability through processes of niche formation: The approach of strategic niche management, Technol. Anal. Strateg. Manage. 10 (2) (1998) 175–198, https://doi.org/10.1080/ 09537329808524310
- [74] B.K. Sovacool, M. Martiskainen, Hot transformations: governing rapid and deep household heating transitions in China, Denmark, Finland and the United

- Kingdom, Energy Policy 139 (2020), 111330, https://doi.org/10.1016/j.
- [75] D. Angel, M.T. Rock, Environmental rationalities and the development state in East Asia: Prospects for a sustainability transition, Technol. Forecast. Soc. Change 76 (2) (2009) 229–240, https://doi.org/10.1016/j.techfore.2008.01.004.
- [76] L.-B. Fischer, J. Newig, Importance of actors and agency in sustainability transitions: a systematic exploration of the literature, Sustainability 8 (5) (2016) 476
- [77] F.W. Geels, Regime resistance against low-carbon transitions: introducing politics and power into the multi-level perspective, Theory Cult. Soc. 31 (5) (2014) 21–40. https://doi.org/10.1177/0263276414531627.
- [78] I. Scoones, The politics of sustainability and development, Annu. Rev. Environ. Resour. 41 (1) (2016) 293–319, https://doi.org/10.1146/annurev-environ-110615-090039
- [79] F.W. Geels, B.K. Sovacool, T. Schwanen, S. Sorrell, Sociotechnical transitions for deep decarbonization, Science 357 (6357) (2017) 1242–1244, https://doi.org/ 10.1126/science.aao3760.
- [80] T. Kalinowski, The politics of climate change in a neo-developmental state: The case of South Korea. International Political Science Review, 2020. 0(0): p. 0192512120924741 DOI: 10.1177/0192512120924741.
- [81] H. Han, Korea's pursuit of low-carbon green growth: a middle-power state's dream of becoming a green pioneer, Pacific Rev. 28 (5) (2015) 731–754, https://doi.org/10.1080/09512748.2015.1013491.
- [82] T. Altenburg, A. Pegels, Sustainability-oriented innovation systems managing the green transformation, Innovat. Dev. 2 (1) (2012) 5–22, https://doi.org/ 10.1080/2157930X.2012.664037.
- [83] S. Jacobsson, A. Bergek, Transforming the energy sector: the evolution of technological systems in renewable energy technology, Indus. Corporate Change 13 (5) (2004) 815–849, https://doi.org/10.1093/icc/dth032.
- [84] World Bank The East Asian Miracle Economic Growth and Public Policy 1993 Oxford University Press Oxford.
- [85] A. Smith, R. Raven, What is protective space? Reconsidering niches in transitions to sustainability, Res. Policy 41 (6) (2012) 1025–1036, https://doi.org/10.1016/ i.respol.2011.12.012.
- [86] S. Steinhilber, P. Wells, S. Thankappan, Socio-technical inertia: Understanding the barriers to electric vehicles, Energy Policy 60 (2013) 531–539, https://doi. org/10.1016/j.enpol.2013.04.076.
- [87] L. Matthews, J. Lynes, M. Riemer, T. Del Matto, N. Cloet, Do we have a car for you? Encouraging the uptake of electric vehicles at point of sale, Energy Policy 100 (2017) 79–88, https://doi.org/10.1016/j.enpol.2016.10.001.
- [88] H.E. Normann, Conditions for the deliberate destabilisation of established industries: Lessons from U.S. tobacco control policy and the closure of Dutch coal mines, Environ. Innovat. Soc. Trans. 33 (2019) 102–114, https://doi.org/ 10.1016/j.eist.2019.03.007.
- [89] R. Eckersley, Greening states and societies: from transitions to great transformations, Environ. Polit. (2020) 1–21, https://doi.org/10.1080/ 09644016.2020.1810890.
- [90] R. MacNeil, Seeding an energy technology revolution in the united states: reconceptualising the nature of innovation in 'liberal-market economies', New Polit. Econ. 18 (1) (2013) 64–88, https://doi.org/10.1080/ 13563467 2012 658362
- [91] Hoogma, R., R. Kemp, J. Schot, and B. Truffer, eds. Experimenting for Sustainable Transport: The approach of Strategic Niche Management. 2002, Routledge: London and New York.
- [92] P. Johnstone, K.S. Rogge, P. Kivimaa, C.F. Fratini, E. Primmer, A. Stirling, Waves of disruption in clean energy transitions: Sociotechnical dimensions of system disruption in Germany and the United Kingdom, Energy Res. Social Sci. 59 (2020), 101287, https://doi.org/10.1016/j.erss.2019.101287.
- [93] J. Fagerberg, Mobilizing innovation for sustainability transitions: a comment on transformative innovation policy, Res. Policy 47 (9) (2018) 1568–1576, https://doi.org/10.1016/j.respol.2018.08.012.
- [94] D. Lang, A. Wiek, M. Bergmann, M. Stauffacher, P. Martens, P. Moll, M. Swilling, C. Thomas, Transdisciplinary research in sustainability science: practice, principles, and challenges, Sustain. Sci. 7 (1) (2012) 25–43, https://doi.org/ 10.1007/s11625-011-0149-x.
- [95] J. Andrews, B. Shabani, The role of hydrogen in a global sustainable energy strategy, Wiley Interdiscipl. Rev. Energy Environ. 3 (5) (2014) 474–489, https://doi.org/10.1002/wene.103.
- [96] Evans, P. and P. Heller, Human Development, State Transformation, and the Politics of the Developmental State, in The Oxford Handbook of Transformations of the State (Oxford Handbooks Online), S. Leibfried, et al., Editors. 2015, Oxford University Press.
- [97] P. Harborne, C. Hendry, Commercialising new energy technologies: failure of the Japanese machine? Technol. Anal. Strateg. Manage. 24 (5) (2012) 497–510, https://doi.org/10.1080/09537325.2012.674671.
- [98] H. Lee, E.-Y. Jung, J.-D. Lee, Public–private co-evolution and niche development by technology transfer: a case study of state-led electricity system transition in South Korea, Energy Res. Social Sci. 49 (2019) 103–113, https://doi.org/ 10.1016/j.erss.2018.11.001.
- [99] S.-Y. Kim, National competitive advantage and energy transitions in Korea and Taiwan, New Polit. Econ. (2020) 1–17, https://doi.org/10.1080/ 13563467.2020.1755245.
- [100] D.P.M. Lam, B. Martín-López, A. Wiek, E.M. Bennett, N. Frantzeskaki, A.I. Horcea-Milcu, D.J. Lang, Scaling the impact of sustainability initiatives: a typology of amplification processes, Urban Transform. 2 (1) (2020) 3, https://doi.org/ 10.1186/s42854-020-00007-9.

- [101] Z. Tzankova, Public policy spillovers from private energy governance: new opportunities for the political acceleration of renewable energy transitions, Energy Res. Soc. Sci. 67 (2020), 101504, https://doi.org/10.1016/j. energy 2020.115544
- [102] C.M.L. Wong, The developmental state in ecological modernization and the politics of environmental framings: the case of singapore and implications for east asia, Nat. Cult. 7 (1) (2012) 95–119, https://doi.org/10.3167/nc.2012.070106.
- [103] L. Hughes, Climate converts: institutional redeployment, industrial policy, and public investment in energy in Japan, J. East Asian Stud. 12 (1) (2012) 89–117, https://doi.org/10.1017/S1598240800007633.
- [104] M. Nilsson, B. Nykvist, Governing the electric vehicle transition Near term interventions to support a green energy economy, Appl. Energy 179 (2016) 1360–1371, https://doi.org/10.1016/j.apenergy.2016.03.056.
- [105] A. Klitkou, S. Bolwig, T. Hansen, N. Wessberg, The role of lock-in mechanisms in transition processes: the case of energy for road transport, Environ. Innovat. Soc. Trans. 16 (2015) 22–37, https://doi.org/10.1016/j.eist.2015.07.005.
- [106] H. Han, Singapore, a garden city: authoritarian environmentalism in a developmental state, J. Environ. Devel. 26 (1) (2017) 3–24, https://doi.org/ 10.1177/1070496516677365
- [107] R. Lowes, B. Woodman, O. Fitch-Roy, Policy change, power and the development of Great Britain's Renewable Heat Incentive, Energy Policy 131 (2019) 410–421, https://doi.org/10.1016/j.enpol.2019.04.041.
- [108] P. Kivimaa, S. Hyysalo, W. Boon, L. Klerkx, M. Martiskainen, J. Schot, Passing the baton: How intermediaries advance sustainability transitions in different phases, Environ. Innovat. Soc. Trans. 31 (2019) 110–125, https://doi.org/10.1016/j. eist.2019.01.001.
- [109] S. Jacobsson, V. Lauber, The politics and policy of energy system transformation—explaining the German diffusion of renewable energy technology, Energy Policy 34 (3) (2006) 256–276, https://doi.org/10.1016/j. enpol.2004.08.029.
- [110] R. Quitzow, Dynamics of a policy-driven market: the co-evolution of technological innovation systems for solar photovoltaics in China and Germany, Environ. Innovat. Soc. Trans. 17 (2015) 126–148, https://doi.org/10.1016/j. eist.2014.12.002.
- [111] P. Cooke, Transition regions: regional–national eco-innovation systems and strategies, Progr. Plann. 76 (3) (2011) 105–146, https://doi.org/10.1016/j. progress.2011.08.002.
- [112] G.C. Chen, C. Lees, Political recentralisation and the diffusion of solar energy in China, Europe-Asia Studies (2019) 1–21, https://doi.org/10.1080/ 09668136.2019.1619669.
- [113] F. Block, Swimming against the current: the rise of a hidden developmental State in the United States, Polit. Soc. 36 (2) (2008) 169–206, https://doi.org/10.1177/ 0032329208318731.
- [114] C. Mazur, M. Contestabile, G.J. Offer, N.P. Brandon, Assessing and comparing German and UK transition policies for electric mobility, Environ. Innovat. Soc. Trans. 14 (2015) 84–100, https://doi.org/10.1016/j.eist.2014.04.005.
- [115] S.V. Valentine, B.K. Sovacool, The socio-political economy of nuclear power development in Japan and South Korea, Energy Policy 38 (12) (2010) 7971–7979. https://doi.org/10.1016/j.enpol.2010.09.036
- 7971–7979, https://doi.org/10.1016/j.enpol.2010.09.036.
   [116] M. Nilsson, K. Hillman, T. Magnusson, How do we govern sustainable innovations? Mapping patterns of governance for biofuels and hybrid-electric vehicle technologies, Environ. Innovat. Soc. Trans. 3 (2012) 50–66, https://doi.org/10.1016/j.eist.2012.04.002.
- [117] E. Figenbaum, Perspectives on Norway's supercharged electric vehicle policy, Environ. Innovat. Soc. Trans. 25 (2017) 14–34, https://doi.org/10.1016/j. eist 2016 11 002
- [118] J.A. Mathews, Green growth strategies—Korean initiatives, Futures 44 (8) (2012) 761–769, https://doi.org/10.1016/j.futures.2012.06.002.
- [119] S.V. Valentine, B.K. Sovacool, Energy transitions and mass publics: manipulating public perception and ideological entrenchment in Japanese nuclear power policy, Renew. Sustain. Energy Rev. 101 (2019) 295–304, https://doi.org/ 10.1016/j.rser.2018.11.008.
- [120] B.D. Leibowicz, Policy recommendations for a transition to sustainable mobility based on historical diffusion dynamics of transport systems, Energy Policy 119 (2018) 357–366, https://doi.org/10.1016/j.enpol.2018.04.066.
- [121] S.-Y. Kim, Hybridized industrial ecosystems and the makings of a new developmental infrastructure in East Asia's green energy sector, Rev. Int. Polit. Econ. 26 (1) (2019) 158–182, https://doi.org/10.1080/ 09692290.2018.1554540.
- [122] P. Kivimaa, E. Primmer, J. Lukkarinen, Intermediating policy for transitions towards net-zero energy buildings, Environ. Innovat. Soc. Trans. 36 (2020) 418–432, https://doi.org/10.1016/j.eist.2020.01.007.
- [123] H. Brauers, P.-Y. Oei, P. Walk, Comparing coal phase-out pathways: the United Kingdom's and Germany's diverging transitions, Environ. Innovat. Soc. Trans. 37 (2020) 238–253, https://doi.org/10.1016/j.eist.2020.09.001.
- [124] P. Johnstone, A. Stirling, B. Sovacool, Policy mixes for incumbency: Exploring the destructive recreation of renewable energy, shale gas 'fracking', and nuclear power in the United Kingdom, Energy Res. Soc. Sci. 33 (2017) 147–162, https:// doi.org/10.1016/j.erss.2017.09.005.
- [125] B.K. Sovacool, B. Turnheim, M. Martiskainen, D. Brown, P. Kivimaa, Guides or gatekeepers? Incumbent-oriented transition intermediaries in a low-carbon era, Energy Res. Soc. Sci. 66 (2020), 101490, https://doi.org/10.1016/j. erss.2020.101490.
- [126] P.A. Hall, D. Soskice, Varieties of Capitalism: The Institutional Foundations of Comparative Advantage, Oxford University Press, United States, 2001.

- [127] B.K. Sovacool, J. Axsen, S. Sorrell, Promoting novelty, rigor, and style in energy social science: Towards codes of practice for appropriate methods and research design, Energy Res. Social Sci. 45 (2018) 12–42, https://doi.org/10.1016/j. erss 2018 07 007
- [128] G. Trencher, A. Taeihagh, M. Yarime, Overcoming barriers to developing and diffusing fuel-cell vehicles: governance strategies and experiences in Japan, Energy Policy 142 (2020), https://doi.org/10.1016/j.enpol.2020.111533.
- [129] Outlook 2020 https://www.iea.org/reports/global-ev-outlook-2020. 2020 [cited 2020 1 December]; Available from.
- [130] IEA, Future of Hydrogen. 2019: Paris.
- [131] G.E. Haslam, J. Jupesta, G. Parayil, Assessing fuel cell vehicle innovation and the role of policy in Japan, Korea, and China, Int. J. Hydrogen Energy 37 (19) (2012) 14612–14623, https://doi.org/10.1016/j.ijhydene.2012.06.112.
- [132] L. Xu, J. Su, From government to market and from producer to consumer: Transition of policy mix towards clean mobility in China, Energy Policy 96 (2016) 328–340, https://doi.org/10.1016/j.enpol.2016.05.038.
- [133] S. Ou, X. Hao, Z. Lin, H. Wang, J. Bouchard, X. He, S. Przesmitzki, Z. Wu, J. Zheng, R. Lv, L. Qi, T.J. LaClair, Light-duty plug-in electric vehicles in China: an overview on the market and its comparisons to the United States, Renew. Sustain. Energy Rev. 112 (2019) 747–761, https://doi.org/10.1016/j.rser.2019.06.021.
- [134] N. Bento, C. Wilson, Measuring the duration of formative phases for energy technologies, Environ. Innovat. Soc. Trans. 21 (2016) 95–112, https://doi.org/ 10.1016/j.eist.2016.04.004.
- [135] L. Kanger, J. Schot, User-made immobilities: a transitions perspective, Mobilities 11 (4) (2016) 598-613, https://doi.org/10.1080/17450101.2016.1211827.
- [136] A. Schaffrin, S. Sewerin, S. Seubert, Toward a comparative measure of climate policy output, Policy Stud. J. 43 (2) (2015) 257–282, https://doi.org/10.1111/ psi 12095
- [137] Y. Wang, D. Sperling, G. Tal, H. Fang, China's electric car surge, Energy Policy 102 (2017) 486–490, https://doi.org/10.1016/j.enpol.2016.12.034.
- [138] Feng, K. and J. Li, Challenges in Reshaping the Sectoral Innovation System of the Chinese Automobile Industry, in Innovation, Economic Development, and Intellectual Property in India and China: Comparing Six Economic Sectors, K.-C. Liu and U.S. Racherla, Editors. 2019, Springer Singapore: Singapore. pp. 415-438.
- [139] J. Li, Charging Chinese future: the roadmap of China's policy for new energy automotive industry, Int. J. Hydrogen Energy 45 (20) (2020) 11409–11423, https://doi.org/10.1016/j.ijhydene.2020.02.075.
- [140] F. Dong, Y. Liu, Policy evolution and effect evaluation of new-energy vehicle industry in China, Resour. Policy 67 (2020), 101655, https://doi.org/10.1016/j. resourpol.2020.101655.
- [141] C. Marquis, H. Zhang, L. Zhou, China's quest to adopt electric vehicles, Stanford Innovat. Rev. 11 (2013) 54–57.
- [142] N. Zhou, Q. Wu, X. Hu, Research on the policy evolution of China's New energy vehicles industry, Sustainability 12 (2020) 3629, https://doi.org/10.3390/ su12093629.
- [143] P. Yu, J. Zhang, D. Yang, S. Lin, T. Xu, The evolution of China's new energy vehicle industry from the perspective of a technology–market–policy framework, Sustainability 11 (2019), https://doi.org/10.3390/su11061711.
- [144] Kennedy, S. China's Risky Drive into New-Energy Vehicles. 2018 [cited 2020 November 18]; Available from: https://www.csis.org/analysis/chinas-risky-dri ve-new-energy-vehicles.
- [145] Y. Liu, A. Kokko, Who does what in China's new energy vehicle industry? Energy Policy 57 (2013) 21–29, https://doi.org/10.1016/j.enpol.2012.05.046.
- [146] Tang, J., China's strategy towards becoming an automotive power in 2030 (Chugoku 2030 Jidoshya kyokoku he no chyosen) In Japanese: . 2019, Tokyo: Nikkei Book.
- [147] X. Hou, P. Li, Whose legitimacy? China's drive for electric vehicles, Sociol. Devel. 6 (1) (2020) 66–90, https://doi.org/10.1525/sod.2020.6.1.66.
- [148] Pearson, M., Local Government and Firm Innovation: China's Clean Energy Sector, in Policy, Regulation and Innovation in Chinese Industry, L. Brandt and T. Rawski, Editors. 2019, Cambridge University Press: New York. p. 96–133.
- [149] Z. Song, Y. Liu, H. Gao, S. Li, The underlying reasons behind the development of public electric buses in China: the Beijing case, Sustainability 12 (2020) 688, https://doi.org/10.3390/su12020688.
- [150] Pelkonen, A. Case Study Report: New Energy Vehicles (China). 2018 [cited 2021 February 2]; Available from: https://jiip.eu/mop/wp/wp-content/uploads/2018/09/CN\_New-Energy-Vehicles\_Pelkonen.pdf.
- [151] D.J. Teece, China and the reshaping of the auto industry: a dynamic capabilities perspective, Manage. Organ. Rev. 15 (1) (2019) 177–199, https://doi.org/ 10.1017/mor.2019.4.
- [152] J. Schwabe, From "obligated embeddedness" to "obligated Chineseness"? Bargaining processes and evolution of international automotive firms in China's New Energy Vehicle sector, Growth Change 51 (3) (2020) 1102–1123, https://doi.org/10.1111/grow.12393.
- [153] Ministry of Economy Trade and Industry (METI). Basic Hydrogen Strategy. 2017 [cited 2018 17 May]; Available from: http://www.meti.go.jp/english/press /2017/1226 003.html.
- [154] Ministry of Economy Trade and Industry (METI). Strategic Roadmap for Hydrogen and Fuel Cells. 2019 [cited 2019 9 August]; Available from: https://www.meti.go.jp/english/press/2019/0312\_002.html.
- [155] Avadikyan, A. and Y. Harayama, The Japanese R&D system in the field of fuel cell vehicles, in The Economic Dynamics of Fuel Cell Technologies, A. Avadikyan, P. Cohendet, and J.-A. Héraud, Editors. 2003, Springer Berlin Heidelberg: Berlin, Heidelberg. p. 187-206.
- [156] Ishitani, H. and Y. Baba, The Japanese strategy for R&D on fuel-cell technology and on-road verification test of fuel-cell vehicles, in Making choices about

- hydrogen: Transport issues for developing countries. 2008, United Nations University Press: New York. p. 64-84.
- [157] Valovirta, V. Case Study Report: Hydrogen Society (Japan). 2018 [cited 2019 10 July]; Available from: http://www.jiip.eu/mop/wp/wp-content/uploads/2018/0 9/JP\_Hydrogen-Society\_Valovirta.pdf.
- [158] K. Hikima, M. Tsujimoto, M. Takeuchi, Y. Kajikawa, Transition analysis of budgetary allocation for projects on hydrogen-related technologies in Japan, Sustainability 12 (20) (2020), https://doi.org/10.3390/su12208546.
- [159] E. Moe, Vested interests, energy efficiency and renewables in Japan, Energy Policy 40 (2012) 260–273, https://doi.org/10.1016/j.enpol.2011.09.070.
- [160] N. Behling, M.C. Williams, S. Managi, Fuel cells and the hydrogen revolution: Analysis of a strategic plan in Japan, Econ. Anal. Policy 48 (2015) 204–221, https://doi.org/10.1016/j.eap.2015.10.002.
- [161] A.C. Lloyd, The california fuel cell partnership: an avenue to clean air, J. Power Sour. 86 (1) (2000) 57–60, https://doi.org/10.1016/S0378-7753(99)00457-7.
- [162] G. Collantes, D. Sperling, The origin of California's zero emission vehicle mandate, Transport. Res. A Policy Pract. 42 (10) (2008) 1302–1313, https://doi. org/10.1016/j.tra.2008.05.007.
- [163] G. Yeung, 'Made in China 2025': the development of a new energy vehicle industry in China, Area Devel. Policy 4 (1) (2019) 39–59, https://doi.org/ 10.1080/23792949.2018.1505433.
- [164] State Council. New Energy Vehicle Industry Development Plan (2021-2035). 2020 [cited 2020 1 December]; Available from: https://www.sustainabletransport.org/archives/7921#\_ftn1.
- [165] W. Li, M. Yang, S. Sandu, Electric vehicles in China: a review of current policies, Energy Environ. 29 (8) (2018) 1512–1524, https://doi.org/10.1177/ 0958305x18781898
- [166] Heller, P. Chinese Government Support for New Energy Vehicles. 2017 [cited 2020 November 21]; Available from: https://www.nbr.org/publication/chinesegovernment-support-for-new-energy-vehicles-as-a-trade-battleground.
- [167] Y. Xue, J. You, X. Liang, H. Liu, Adopting strategic niche management to evaluate EV demonstration projects in China, Sustainability 8 (2016), https://doi.org/ 10.3390/su8020142.
- [168] The Economist. It's the system, stupid China's plans for the electrified, autonomous and shared future of the car. 2019 [cited 2020 November 2020]; Available from: https://www.economist.com/briefing/2019/04/04/chinas-pl ans-for-the-electrified-autonomous-and-shared-future-of-the-car.
- [169] S.-C. Ma, Y. Fan, L. Feng, An evaluation of government incentives for new energy vehicles in China focusing on vehicle purchasing restrictions, Energy Policy 110 (2017) 609–618, https://doi.org/10.1016/j.enpol.2017.07.057.
- [170] State Council. Green auto measures rolled out. 2019 [cited 2020 1 December]; Available from: http://english.www.gov.cn/policies/policywatch/201908/23/content WS5d5f36f1c6d0c6695ff7f2b2.html.
- [171] Y. Li, C. Zhan, M. de Jong, Z. Lukszo, Business innovation and government regulation for the promotion of electric vehicle use: lessons from Shenzhen, China, J. Clean. Product. 134 (2016) 371–383, https://doi.org/10.1016/j. iclepro.2015.10.013.
- [172] J.P. Helveston, Y. Wang, V.J. Karplus, E.R.H. Fuchs, Institutional complementarities: The origins of experimentation in China's plug-in electric vehicle industry, Res. Policy 48 (1) (2019) 206–222, https://doi.org/10.1016/j. respol 2018 08 006
- [173] Tagscherer, U., Electric mobility in China A policy review. 2012, Fraunhofer Institute for Systems and Innovation Research (ISI).
- [174] Yang, J. How China plans to create giant state-owned carmaker. 2017 [cited 2021 January 30]; Available from: www.autonews.com/article/20171217/GLOBAL 03/171219798/how-china-plans-to-create-giant-state-owned-carmaker.
- [175] H. Gong, M.Q. Wang, H. Wang, New energy vehicles in China: policies, demonstration, and progress, Mitig. Adapt. Strat. Glob. Change 18 (2) (2013) 207–228, https://doi.org/10.1007/s11027-012-9358-6.
- [176] Z. Wan, D. Sperling, Y. Wang, China's electric car frustrations, Transport. Res. Part D Trans. Environ. 34 (2015) 116–121, https://doi.org/10.1016/j. trd.2014.10.014.
- [177] Jin, L., Literature review of electric vehicle consumer awareness and outreach activities. 2017, International Council on Clean Transportation.
- [178] Shanghai Municipal People's Government. Auto cluster aims for further upgrades. 2020 [cited 2021 20 February]; Available from: http://www.shanghai.gov. cn/nw48088/20201127/c6a4f12a69e943ebb169a44f5e01ee4a.html.
- [179] Z. Ji, X. Huang, Plug-in electric vehicle charging infrastructure deployment of China towards 2020: Policies, methodologies, and challenges, Renew. Sustain. Energy Rev. 90 (2018) 710–727, https://doi.org/10.1016/j.rser.2018.04.011.
- 180] Hove, A. and A. Sandalow. Electric Vehicle Charging in China and the United States. 2019 [cited 2021 21 January]; Available from: https://www.energypolicy. columbia.edu/sites/default/files/file-uploads/EV\_ChargingChina-CGEP\_Report\_ Final.pdf.
- [181] OECD. Case Study on System Innovation in China The Case of Electric Vehicles. 2014 [cited 2021 2 February]; Available from: http://www.innovationpolicyplat form.org/www.innovationpolicyplatform.org/system/files/CHINA%20-%20The %20Case%20of%20Electric%20Vehicles-%20IPP\_0/index.pdf.
- [182] X. Zhang, R. Rao, J. Xie, Y. Liang, The current dilemma and future path of China's electric vehicle, Sustainability 6 (3) (2014) 1567–1593, https://doi.org/10.3390/ ps/031567.
- [183] Ao, S. Key Insights on China's Policies on Developing New Energy Vehicles. 2020 [cited 2021 February 2]; Available from: https://www.integral-japan.net/? p=3088
- [184] S. Howell Leapfrogging or Stalling Out? Harvard Kennedy School of Governance Electric Vehicles in China. 2014.

- [185] HySTRA. HySTRA CO2-free Hydrogen Energy Supply-chain Technology Research Association. 2019 [cited 2019 August 11]; Available from: http://www. hystra.or.in/en/
- [186] Arias, J. Hydrogen and Fuel Cells in Japan. 2019 [cited 2020 19 October]; Available from: https://www.eu-japan.eu/sites/default/files/publications/docs/hydrogen\_and\_fuel\_cells\_in\_japan.pdf.
- [187] Japan H2 Mobility. About JHyM. 2020 [cited 2020 1 December]; Available from: https://www.jhym.co.jp/en/nav-about.
- [188] S. Iida, K. Sakata, Hydrogen technologies and developments in Japan, Clean Energy 3 (2) (2019) 105–113, https://doi.org/10.1093/ce/zkz003.
- [189] Nagashima, M. Japan's Hydrogen Strategy and it's Economic and Geopolitical Implications. 2018 [cited 2019 15 June]; Available from: https://www.ifri.or g/en/publications/etudes-de-lifri/japans-hydrogen-strategy-and-its-econo mic-and-geopolitical-implications.
- [190] Okutsu, A. and N. Shibata. Be water: Japan's big, lonely bet on hydrogen. 2020 [cited 2020 December 30]; Available from: https://asia.nikkei.com/Spotlight/The-Big-Story/Be-water-Japan-s-big-lonely-bet-on-hydrogen.
- [191] P. Harborne, C. Hendry, J. Brown, The development and diffusion of radical technological innovation: the role of bus demonstration projects in commercializing fuel cell technology, Technol. Anal. Strateg. Manage. 19 (2) (2007) 167–188, https://doi.org/10.1080/09537320601168060.
- [192] Ministry of Economy Trade and Industry (METI). World's Largest-Class Hydrogen Producing Base Using Renewable Energy to Start in Namie Town, Fukushima Prefecture. 2020 [cited 2021 January 3]; Available from: https://www.meti.go. jp/english/press/2020/0213\_001.html.
- [193] K. Nakui, An overview of the fuel cell and hydrogen technology development policies in Japan, J. Chem. Eng. Jpn. 39 (5) (2006) 489–502, https://doi.org/ 10.1252/jcei.39.489
- [194] Fuel Cell Commercialization Conference of Japan (FCCJ). Toward Commercialization and Wide Use of fuel cells. 2018 [cited 2020 December 1]; Available from: http://fccj.jp/pdf/fccj\_pamphlet.pdf.
- [195] Association of Hydrogen Supply/Utilization Technology (HySUT). The Association of Hydrogen Supply/Utilization Technology. 2020 [cited 2020 1 December]; Available from: http://hysut.or.jp/en/pdf/pamphlet.pdf.
- [196] M. Åhman, Government policy and the development of electric vehicles in Japan, Energy Policy 34 (4) (2006) 433–443, https://doi.org/10.1016/j. enpol.2004.06.011.
- [197] Hydrogen and Fuel Cell Strategies Council. Suiso Nenryodenchi Senryaku
  Rodomappu no Tassei ni Muketa Taiojokyo (Status of measures to attaining
  hydrogen and fuel cell roadmap) In Japanese:. 2020 [cited 2020 15 December];
  Available from: https://www.meti.go.jp/shingikai/energy\_environmen
  t/suiso nenryo/roadmap hyoka wg/pdf/002 01 00.pdf.
- [198] Toyota. Toyota-Hino Fuel Cell Bus to Serve Centrair and Vicinity. 2006 [cited 2020 December 1]; Available from: https://global.toyota/en/detail/275012.
- [199] Greene, D. Status and Prospects of the Global Automotive Fuel Cell Industry and Plans for Deployment of Fuel Cell Vehicles and Hydrogen Refueling Infrastructure. 2013 [cited 2020 2 December]; Available from: https://www.ener gy.gov/eere/fuelcells/downloads/status-and-prospects-global-automotivefuel-cell-industry-and-plans.
- [200] Yamanashi, U.o. Fuel Cell Nanomaterials Center University of Yamanashi. 2016 [cited 2020 2 December]; Available from: https://fc-nano.yamanashi.ac.jp/english/img/nonoen2016.pdf.
- [201] Nagashima, M. Japan's Hydrogen Society Ambition 2020 Status and Perspectives. 2020 [cited 2020 1 December]; Available from: https://www.ifri.org/en/publications/notes-de-lifri/japans-hydrogen-society-ambition-2020-status-and-perspectives.
- [202] Toyota. Seven-Eleven Japan and Toyota to Launch Joint Next-generation Convenience Store Project in Autumn 2019 toward Greater CO2 Emissions Reduction. 2018 [cited 2020 December 1]; Available from: https://global.toyota/en/newsroom/corporate/22833613.html.
- [203] Ito, H. and Y. Noguchi. Japan to ban domestic gas-fueled car sales by mid-2030s. 2020 [cited 2021 March 1]; Available from: http://www.asahi.com/ajw/articles/13989088#:--text=Japan%20will%20ban%20the%20sale,zero%20carbon%20emissions%20bv%202050.
- [204] CaFCP. The California Fuel Cell Revolution: A Vision for Advancing Economic, Social and Environmental Priorities. 2018 [cited 2020 April 30]; Available from: https://cafcp.org/sites/default/files/CAFCR.pdf.
- [205] California Air Resources Board (CARB). 2020 Annual Evaluation of Fuel Cell Electric Vehicle Deployment & Hydrogen Fuel Station Network Development. 2020 [cited 2020 December 5]; Available from: https://ww2.arb.ca.gov/sites/default/files/2020-09/ab8\_report\_2020.pdf.
- [206] CEC & CARB. Joint Agency Staff Report on Assembly Bill 8: 2019 Annual Assessment of Time and Cost Needed to Attain 100 Hydrogen Refueling Stations in California. 2019 [cited 2020 5 March]; Available from: https://ww2.energy.ca. gov/2019publications/CEC-600-2019-039/CEC-600-2019-039.pdf.
- [207] Governor of the State of California. ZEV Action Plan: A roadmap toward 1.5 million zero-emission vehicles on California roadways by 2025. 2013 [cited 2021 10 January]; Available from: http://opr.ca.gov/docs/Governors\_Office\_ZEV\_Action\_Plan\_(02-13).pdf.
- [208] Governor of the State of California. ZEV Action Plan: Priorities Update. 2018 [cited 2021 10 January]; Available from: https://static.business.ca.gov/wp-content/uploads/2019/12/2018-ZEV-Action-Plan-Priorities-Update.pdf.
- [209] McConnell, V., B. Leard, and F. Kardos. California's Evolving Zero Emission Vehicle Program: Pulling New Technology into the Market. 2019 [cited 2020 10 May]; Available from: https://media.rff.org/documents/RFF\_WP\_Californias\_Evolving\_Zero\_Emission\_Vehicle\_Program.pdf.

- [210] J.H. Wesseling, J.C.M. Farla, D. Sperling, M.P. Hekkert, Car manufacturers' changing political strategies on the ZEV mandate, Transport. Res. D Transport Environ. 33 (2014) 196–209, https://doi.org/10.1016/j.trd.2014.06.006.
- [211] Advanced Clean Trucks Fact Sheet. 2020 https://ww2.arb.ca.gov/resources/fact-sheets/advanced-clean-trucks-fact-sheet. [cited 2021 January 20]; Available from
- [212] California Air Resources Board. Low Carbon Fuel Standard: About. No date [cited 2021 January 30]; Available from: ww2.arb.ca.gov/our-work/programs/lowcarbon-fuel-standard/about.
- [213] California Air Resources Board. Proposed Fiscal Year 2019-20 Funding Plan for Clean Transportation Incentives. 2019 [cited 2021 12 February]; Available from: https://www2.arb.ca.gov/sites/default/files/2019-09/fy1920fundingplan.pdf.
- [214] California Air Resources Board. Proposed Fiscal Year 2020-21 Funding Plan for Clean Transportation Incentives. 2020 [cited 2021 12 February]; Available from: https://ww2.arb.ca.gov/sites/default/files/2020-11/proposed\_fy2020-21\_fundingplan.pdf.
- [215] California Fuel Cell Partnership. Members. 2021 [cited 2021 January 20]; Available from: https://cafcp.org/members.
- [216] California Fuel Cell Partnership. About us. 2020 [cited 2020 5 December]; Available from: https://cafcp.org/about\_us.
- [217] California Hydrogen Business Council. 2012 Annual Report. 2013 [cited 2020 5 December]; Available from: http://www.californiahydrogen.org/wp-content/uploads/2017/11/Annual-Report-2012.pdf.
- [218] California Hydrogen Business Council. 2019 Annual Report. 2020 [cited 2020 5 December]; Available from: https://www.californiahydrogen.org/wp-conten t/uploads/2020/07/Annual-Report-2019-Draft.pdf.
- [219] California Fuel Cell Partnership. A California Road Map: Bringing Hydrogen Fuel Cell Electric Vehicles to the Golden State. 2012 [cited 2021 January 2021]; Available from: https://cafcp.org/sites/default/files/20120814\_Roadmapv% 28Overview%29.pdf.
- [220] California Air Resources Board (CARB). 2019 Annual Evaluation of Fuel Cell Electric Vehicle Deployment & Hydrogen Fuel Station Network Development. 2019 [cited 2020 March 5]; Available from: https://ww2.arb.ca.gov/sites/default/files/2019-07/AB8 report 2019 Final.pdf.
- [221] T.E. Lipman, A.L. Gray-Stewart, J. Lidicker, Driver response to hydrogen fuel cell buses in a real-world setting: study of a northern california transit bus fleet, Transp. Res. Rec. 2502 (1) (2015) 48–52, https://doi.org/10.3141/2502-06.
- [222] T. Lipman, M. Witt, M. Elke, Lessons learned from the installation and operation of Northern California's first 70-MPa hydrogen fueling station, Int. J. Hydrogen Energy 38 (36) (2013) 15868–15877, https://doi.org/10.1016/j. iihydene.2013.08.120.
- [223] California Air Resources Board (CARB). Annual Evaluation of Fuel Cell Electric Vehicle Deployment & Hydrogen Fuel Station Network Development. 2014 [cited 2020 March 5]; Available from: https://ww2.arb.ca.gov/sites/default/files/2020 -10/ab8 report final june2014 ac.pdf.
- [224] Toyota. The Future of Zero-Emission Trucking Takes Another Leap Forward (Press Release). 2019 [cited 2020 10 December]; Available from: https://pressroom. toyota.com/the-future-of-zero-emission-trucking-takes-another-leap-forward/.
- [225] Eudy, L., K. Chandler, and C. Gikakis. Fuel Cell Buses in U.S. Transit Fleets: Summary of Experiences and Current Status. 2007 [cited 2020 12 December]; Available from: https://www.energy.gov/sites/prod/files/2014/03/f12/41967. pdf.
- [226] California Fuel Cell Partnership. Fuel Cell Electric Buses: Enable 100% Zero Emission Bus Procurement by 2029. 2019 [cited 2020 April 20]; Available from: https://cafcp.org/sites/default/files/2019-CaFCP-FCEB-Road-Map.pdf.
- [227] California Air Resources Board. Annual Evaluation of Fuel Cell Electric Vehicle Deployment and Hydrogen Fuel Station Network Development. 2014 [cited 2021 January 10]; Available from: https://ww2.arb.ca.gov/sites/default/files/2020-10/ab8 report final june2014\_ac.pdf.
- [228] California Fuel Cell Partnership. A Road Map for Fuel Cell Electric Buses in California. 2013 [cited 2020 December 12]; Available from: https://cafcp.org/sit es/default/files/A%20Roadmap%20for%20Fuel%20Cell%20Electric%20Buses% 20in%20California.pdf.
- [229] Issues. 2021 https://www.californiahydrogen.org/policy-issues. [cited 2021 January 10]; Available from.
- [230] California Fuel Cell Partnership. Medium and Heavy-Duty Fuel Cell Electric Truck Action Plan for California. 2016 [cited 2020 April 20]; Available from: https://cafcp.org/mdhd-action-plan-2016.
- [231] California Energy Commission and California Air Resources Board. Joint Agency Staff Report on Assembly Bill 8: Assessment of Time and Cost Needed to Attain 100 Hydrogen Refueling Stations in California. 2015 [cited 2021 January 20]; Available from: https://cafcp.org/sites/default/files/CEC\_ARB\_Joint\_Staff\_R
- [232] California Air Resources Board (CARB). 2018 Annual Evaluation of Fuel Cell Electric Vehicle Deployment & Hydrogen Fuel Station Network Development. 2018 [cited 2020 March 22]; Available from: https://ww2.arb.ca.gov/sites/default/files/2018-12/ab8\_report\_2018\_print.pdf.
- [233] Honda Newsroom. Honda Supporting Growth of California Hydrogen Network with Financial Support to FirstElement Fuel. 2014 [cited 2021 January 20]; Available from: https://hondanews.com/en-US/releases/honda-supporting-gr owth-of-california-hydrogen-network-with-financial-support-to-firstelement-fuel.
- [234] California Energy Commission. 2013-2014 Investment Plan Update for the Alternative and Renewable Fuel and Vehicle Technology Program. 2013 [cited 2021 January 22]; Available from: https://www2.energy.ca.gov/publications/dis playOneReport\_cms.php?pubNum=CEC-600-2012-008-CMF.

- [235] California Energy Commission. 2017-2018 Investment Plan Update for the Alternative and Renewable Fuel and Vehicle Technology Program. 2017 [cited 2021 January 22]; Available from: https://www.energy.ca.gov/proceeding s/energy-commission-proceedings/inactive-proceedings/2017-2018-investmen t-plan-proceeding.
- [236] California Air Resources Board. Innovative Clean Transit (ICT) Regulation Fact Sheet. 2019; Available from: https://ww2.arb.ca.gov/resources/fact-sheets/inn ovative-clean-transit-ict-regulation-fact-sheet.
- [237] California Air Resources Board. Appendix B: CARB's Zero Emission Vehicle Programs. 2019 [cited 2021 February 2]; Available from: https://ww2.arb.ca.
- gov/sites/default/files/2019-12/SB%20498%20Appendix%20B%20-%20ZEV%20Programs%20120719.pdf.
- [238] Matsuoka, D. Toyota president criticizes Japan's swift move to phase out gasoline cars. 2020 [cited 2021 February 20]; Available from: https://mainichi.jp/ english/articles/20201218/p2a/00m/0bu/004000c.
- [239] G. Trencher, J, Van der heijden, Instrument interactions and relationships in policy mixes: Achieving complementarity in building energy efficiency policies in New York, Sydney and Tokyo, Energy Res. Soc. Sci. 54 (2019) 34–45, https://doi.org/ 10.1016/j.erss.2019.02.023.
- [240] A. Taeihagh, Network-centric policy design, Policy Sci. (2017) 1–22, https://doi. org/10.1007/s11077-016-9270-0.